

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Miglo, Anton

# Article Crowdfunding in a competitive environment

Journal of Risk and Financial Management

**Provided in Cooperation with:** MDPI – Multidisciplinary Digital Publishing Institute, Basel

*Suggested Citation:* Miglo, Anton (2020) : Crowdfunding in a competitive environment, Journal of Risk and Financial Management, ISSN 1911-8074, MDPI, Basel, Vol. 13, Iss. 3, pp. 1-38, https://doi.org/10.3390/jrfm13030039

This Version is available at: https://hdl.handle.net/10419/239141

#### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.







# Article Crowdfunding in a Competitive Environment <sup>+</sup>

# Anton Miglo

Department of Finance, Accountancy and Economics, Birmingham City University, Birmingham B4 7XG, UK; anton.miglo@bcu.ac.uk

+ This paper initially circulated as "Crowdfunding in a Duopoly Under Asymmetric Information". I would like to thank professor Douglas Cumming for his encouragement. Also many thanks to Giacomo Corneo, Julie Ann Elston, Kai Hueschelrath, William E. Jackson III, Xiehua (Richard) Ji, Victor Miglo, Juuso Valimaki, two anonymous referees and the seminar participants at University of Brighton, de Montfort University, University of Greenwich, University of Westminster, London South Bank University, the Royal Economic Society annual meeting of 2018 and the British Accounting and Finance Association annual meeting of 2018 for the useful comments and editing assistance.

Received: 14 December 2019; Accepted: 20 February 2020; Published: 25 February 2020



**Abstract:** Crowdfunding has mostly been used to finance very unique projects. Recently, however, companies have begun using it to finance more traditional products where they compete against other sellers of similar products. Major crowdfunding platforms, Kickstarter and Indiegogo, as well as Amazon have launched several projects consistent with this trend. This paper offers a model where two competing firms can use crowdfunding prior to direct sales. The model provides several implications that have not yet been tested e.g., (1) Firms can use crowdfunding strategically to signal a high level of demand for their products; (2) (Reward-based) crowdfunding is procyclical; (3) A higher platform fee may lead to higher firm profits in equilibrium; (4) Competition increases the chances of using crowdfunding compared to the monopoly case; (5) A non-monotonic relationship exists between the risk of crowdfunding campaign failure and firm profit.

**Keywords:** asymmetric information; crowdfunding and competition; reward-based crowdfunding; signalling; strategic entrepreneurship

JEL Codes: D43; D82; G32; L11; L13; L26; M13

# 1. Introduction

Crowdfunding is a form of fundraising whereby groups of people pool money, typically (very) small individual contributions, to support a particular goal (Ahlers et al. (2015)). It is usually performed online. There are different types of crowdfunding. Our focus in this paper is on reward-based crowdfunding,<sup>1</sup> where investors count on some extra-benefits from the company such as future product discounts. Reward-based crowdfunding campaigns are commonly offered in one of two models. The "All-Or-Nothing" (AON) model involves establishing a fundraising goal. If the goal is not achieved, the money are returned to investors/funders. The "Keep-It-All" (KIA) model assumes keeping the entire amount raised unconditionally.

The two largest crowdfunding platforms are Kickstarter and Indiegogo. Kickstarter follows an AON model. Since its creation in 2008, it hosted 473,941 campaigns, among which 176,497 successfully funded for a total amount raised of more than \$4.5 billion.<sup>2</sup> Indiegogo is the second largest

Other types are equity-based crowdfunding, debt-based crowdfunding and donation-based crowdfunding. We will discuss them in Section 7

<sup>&</sup>lt;sup>2</sup> https://www.kickstarter.com/help/stats, retrieved 08/02/2020.

crowdfunding platform. It hosted more than 800,000 campaigns, among which about 9% successfully funded for a total amount raised of more than \$1.6 billion (Smith  $(2020)^3$ ). On Indiegogo entrepreneurs can choose between a KIA model and an AON model. The platforms charge fees that can range from 3% to 9% of the volume of funds raised.

Compared to Kickstarter, Indiegogo is a more open platform that can be used by any company for any product.<sup>4</sup> It launched several strategic projects in 2017–2018 helping crowdfunding firms develop their business after a campaign. For example, it introduced Indiegogo Marketplace/InDemand where firms sell their products initially financed through crowdfunding and compete against other firms which may not have necessarily used crowdfunding to launch their products (Schleifer (2017); Rear (2018)). In a similar spirit, both Kickstarter and Indiegogo have teamed up with Amazon to create an Amazon Launchpad project. It makes it easy for startups to launch, market, and distribute their products to millions of Amazon customers around the globe. The program offers a number of services for entrepreneurs such as custom product pages and comprehensive marketing packages. It helps firms overcome many of the problems associated with starting new businesses. Both Marketplace and Amazon Launchpad help creating an environment where firms that have used crowdfunding compete against firms that have not. Among other examples note that Australian farmers seem to have started using crowdfunding to improve traditional market structures like duopolies which are often the case (Askew (2016)).

Several interesting features of the previously discussed trends seem to have emerged. First, platform fees related to the projects described above are relatively high. For example, Indiegogo takes 10%-15% of every sale made through Marketplace (Mogg (2017)). Secondly, launching a crowdfunding campaign prior to spot sales has become a part of the strategy for not only "unknown" start-up firms but also for established, reputable firms. David Mandelbrot, CEO of Indiegogo, said the following (Takahashi (2018)): "The other trend we're seeing is more and more companies using Indiegogo as a way to engage with their audience early. In the last year we've had campaigns from companies like Procter and Gamble, Honeywell, and Bose. They're big, public companies...but they're using Indiegogo to validate the products coming out of their... divisions and launch those products to an audience they can engage with directly."

Moreover one can now see cases where products offered by firms who used InDemand or AmazonLaunchpad and had crowdfunding experience previously seem to have a higher quality compared to firms that did not use either. For example the company Prynt, founded in 2014, raised more than \$1.5 million from more than 9000 backers on Kickstarter before launching the The Prynt Case on Amazon Launchpad, with customers already purchasing enough cases and photo packs to print more than 120,000 instant photos.<sup>5</sup> Prior to launching their crowdfunding campaign, its founders had a good level of belief in the high potential of their product. After showing off their product in early 2016 at the Consumer Electronics Show in Las Vegas, Prynt had a couple of notable write-ups under their belt. "...a [Kickstarter] campaign requires a lot of anticipation and dedication...," said Clément Perrot, CEO and Co-Founder of Prynt (Cunningham (2016)). He also said that before you hit the launch button, you have to be ready with updates, posts, pictures and images. Shortly after their crowdfunding campaign, an independent author covering innovations mentioned: "...there are a whole slew of services that let you order prints right from your smartphone with incredible ease and equally incredible quality. There are even apps that channel the wind-and-click days of only getting one shot without any edits. However, compared to Prynt, those innovations start to look like novelty gimmicks." (Cunningham (2016)). Furthermore it seems like on Amazon the ratings, prices and customer feedback of Prynt are relatively high, and higher for example than that those of

<sup>&</sup>lt;sup>3</sup> See also https://enventyspartners.com/blog/kickstarter-vs-indiegogo-platform-choose/

<sup>&</sup>lt;sup>4</sup> https://artofthekickstart.com/kickstarter-vs-indiegogo-and-how-to-decide-for-your-crowdfunding-campaign/

<sup>&</sup>lt;sup>5</sup> https://press.aboutamazon.com/news-releases/news-release-details/amazon-launchpad-announces-kickstartercollection

firms that did not use crowdfunding.<sup>6</sup> "Participating in the Amazon Launchpad program has been wonderful—from their awesome team to all of the resources they make easily available to startups," Perrot said in another interview.<sup>7</sup> "The marketing support in particular is huge in helping us stand out, and our sales have been excellent. It's great to know that we have a team at Amazon focused on making sure our product is a success!" Recently an expert company specializing in rankings on portable printer named Prynt as the best printer in the iPhone category: "…we've chosen to highlight the iPhone-only Prynt Pocket due to its novel and highly-portable design that sets it apart from most other smartphone printers out there..." (Hesse (2019)).

One should also mention the Kickstarter Collection project related to Amazon Launchpad. It features products and projects that had successful crowdfunding experiences with Kickstarter and are now available for purchase on Amazon.com. "...Working with Kickstarter is a great way for us to hear directly from customers what products they care about, since they truly hold the power to bring these products to life...," said Jim Adkins, Vice President of Amazon.<sup>8</sup> Companies in the Kickstarter Collection include Piper, Zivix, MudWatt and the above mentioned Prynt among others. Most of these companies are very successful. Note for example Piper's award winning product "The Piper Computer Kit".<sup>9</sup>

In our model, a firm decides whether to use crowdfunding for the pre-sale stage or just use spot price sales. The crowdfunding campaign has the following features: (1) no arbitrage condition; in equilibrium, the crowdfunding pre-sale price and the spot price are equal; (2) the firm provides rewards to funders; (3) the crowdfunding decision is publicly observable.<sup>10</sup> We first demonstrate that in a monopoly setting, without making any additional assumptions such as community benefits to funders, crowdfunding is never used and firms prefer to just use the spot market since crowdfunding involves a cost in the form of rewards. Then we consider a duopoly. When information between firms is symmetric a situation where both duopolists do not use crowdfunding, and only use the spot market, is not an equilibrium because each firm has an incentive to deviate and use a crowdfunding campaign prior to its spot sales. Early commitment of a firm increases its production, makes the product price lower and forces the other firm out of its optimal quantity. When information between firms is asymmetric, we find that the only signalling equilibrium that exists is when high-quality (high demand) firms are likely to chose reward-based crowdfunding strategically as a signal of quality. Low-quality firms are less likely to mimic high-quality firms (by chosing crowdfunding), as this would imply an additional cost related to rewards, and prefer spot sales exclusively instead. On the other hand an equilibrium where low-quality firms select crowdfunding and high-quality firms do not does not exist. If the uninformed competitor perceives the firm that uses a crowdfunding campaign as a low-quality/low-demand firm, it will be pessimistic about the price and will concede the market. A high-quality firm can benefit from this situation by mimicking the strategy of low-quality firms during the crowdfunding stage. The existence and uniqueness of the signalling equilibrium allows us to generate empirical predictions about the connections between crowdfunding and firm quality/demand.

<sup>&</sup>lt;sup>6</sup> https://www.amazon.com/Prynt-Pocket-Instant-Printer-iPhone/dp/B06XSDSZ6J/ref=dp\_ob\_title\_wld; https://www.amazon.com/KiiPix-Smartphone-Picture-Printer-Blue/product-reviews/B07B7PQGBF?pageNumber=2

<sup>&</sup>lt;sup>7</sup> https://press.aboutamazon.com/news-releases/news-release-details/amazon-launchpad-announces-kickstartercollection

<sup>8</sup> https://press.aboutamazon.com/news-releases/news-release-details/amazon-launchpad-announces-kickstartercollection

<sup>&</sup>lt;sup>9</sup> https://www.playpiper.com/

<sup>&</sup>lt;sup>10</sup> The fact that the crowdfunding decision is well observed publicly, given the nature of crowdfunding where firms use public websites (platforms) to conduct crowdfunding campaigns, makes crowdfunding different from other pre-sale methods such as forward sales for example. Although there is literature that argues that the disclosure of these contracts is desirable (see, among others, Hughes and Kao (1997) and Allaz (1992)), this issue remains quite ambiguous. As another example note that the development of a product within a private company that uses private financing is not as transparent as it would be with crowdfunding etc.

The model provides several implications that have not yet been tested e.g., firms can use crowdfunding strategically to signal a high level of demand for their products; (reward-based) crowdfunding is procyclical; a higher platform fee may lead to higher firm profits in equilibrium; competition increases the chances of using crowdfunding compared to the monopoly case; a non-monotonic relationship exists between the risk of crowdfunding campaign failure and firm profit.

The rest of the paper is organized as follows. Section 2 presents the review of related lietarture. Section 3 presents the basic model and some preliminary analysis. Section 4 analyzes a model with competition. Section 5 provides the main results of the duopoly analysis under asymmetric information. Section 6 summarizes the model's results and predictions. Section 7 discusses the model's robustness and its potential extensions and Section 8 provides a conclusion to the study.

#### 2. Literature Review

Crowdfunding research is quickly growing. Moritz and Block (2014); Kuppuswamy and Bayus (2015); Alegre and Moleskis (2016); Cumming and Johan (2017); Cumming and Hornuf (2018); Estrin et al. (2018); Mochkabadi and Volkmann (2018) and Cumming et al. (2019) provide good reviews of the literature in this field. Several topics seem to emerge that are closely related to the present paper.

Petruzzelli et al. (2019) identify five aspects of research on crowdfunding: platform, backers, campaign, entrepreneurs and outcome. In "campaign" area they point out the importance of the problem of choosing the right campaign type. Similarly, Miglo and Miglo (2019) mentioned that the literature on crowdfunding still lacks a full understanding of how entrepreneurs choose between different types of crowdfunding and how they decide whether to use crowdfunding or other types of financing. Among papers in this area note the following. Belleflamme et al. (2014) compare reward-based and equity-based crowdfunding. As the authors mentioned, further research is required. Miglo and Miglo (2019) consider the choice between the different types of crowdfunding and traditional financing under different types of market imperfections. Fairchild et al. (2017) analyze an entrepreneur's choice between venture capital financing and crowdfunding in order to finance a new innovative start-up. Petruzzelli et al. (2019) suggest several propostions regarding the choice between sustainability-driven crowdfunding campaigns (mostly donations-based crowdfunding) and commercial-based campaigns.

In the present paper we analyze the choice between crowdfunding and spot sales, which has not been analyzed previously.

Some papers analyze links between the crowdfunding campaign and the aspects of firm activity beyond the campaign itself. One line of research looks at this connection from funders/backers point of view. For example, Gleasure and Feller (2016) suggest a "pay-to-partcipate" concept. This concept describes a desire to be part of the community responsible for some project. Similarly, Gerber et al. (2012) noted backers' desires to "engage and contribute to a trusting and creative community". Beaulieu et al. (2015) suggest that crowdfunding is a form of relationship building because of a desire to construct shared meaning. Zvilichovsky et al. (2013) noted the reciprocal social capital created by this relationship. Gambardella (2012) suggest that in general investors may wish to participate because they wish to learn new knowledge or skills and enjoy the opportunity to observe a project unfold. In a similar spirit the concept of erosion of organisations' financial boundaries focuses on funders' expectations to access firm records, digital resources, and other valuable information. Gambardella (2012) provide some examples related to this concept (see also Kuo and Gerber (2012); Frydrych et al. (2014); Mollick (2014); Gleasure (2015) and Chaney (2019)).

A connected line of research looks at links between crowdfunding campaign and "after-campaign" scenarios from the firm/entrepreneur point of view. For example Cumming et al. (2019) point to the importance of determining success chances of firm projects beyond the crowdfunding campaign success. Mollick (2014) reports that 75% of projects successfully funded on Kickstarter deliver late. Cumming et al. (2019) suggest that one might expect that this percentage varies according to the fundraising model (KIA versus AON) used during the campaign, since the latter is related to the

amount raised. Projects that are started with sufficient funds are more likely to produce the promised product and eventually deliver on time, something that is worthwhile investigating in future research. Belleflamme et al. (2014) argued that crowdfunding allowed a form of price discrimination between pre-ordering consumers (crowdfunders) and other consumers that could mitigate risk and up-front costs, provided fundraising did not exceed the threshold at which it cannibalises post-production sales. Similar observations were made by authors such as Singer et al. (2011); Gerber et al. (2012); and Sung-Min (2012), who highlighted the benefits of crowdfunding in allowing creators to engage with potential users earlier in development during a fund-seeking campaign, such as additional opportunities for feedback and experimentation.

Our paper mostly contributes to crowdfunding literature from the previous paragrapth namely by looking at the benefits of crowdfunding for a production and spot sales following a crowdfunding campaign from an entrepreneur's point of view. For example, we highlight the strategic role of crowdfunding as a marketing tool and a signalling device for a firm when it competes against other firms and how the usage of crowdfunding shapes the outcome of a competition between rivals not only in crowdfunding itself but in production and spot sales of products in an "after-campaign" state.

Some papers analyze the role of information in crowdfunding. One line of research focuses on the role of asymmetric information between founders and funders. Although most researchers agree that crowdfunding can play an important role in markets with asymmetric information (see, among others, Ahlers et al. (2015); Kleinert et al. (2020); Hildebrand et al. (2016); Vismara (2018a) and Vismara (2018b), researchers often disagree on the exact nature of this role, e.g., what type of crowdfunding can be used as a signal of a firm's quality, what signals are sent to potential investors if a firm selects crowdfunding instead of the other types of financing etc. Belleflamme et al. (2014) argue that the value of equity-based crowdfunding can be higher under asymmetric information than under symmetric information. Miglo and Miglo (2019) consider the choice between reward-based crowdfunding and equity-based crowdfunding and find that high-quality firms prefer reward-based crowdfunding and use it as a signal of quality. Chakraborty and Swinney (2019) consider a crowdfunding model where product quality is known to the entrepreneur but not to some contributors. They find that a larger campaign target can be used by high-quality firms as a signalling device. At the same time, Miglo and Miglo (2019) find that the relationship between a firm's quality and the campaign goal is non-linear. More specifically they argue that the threshold should be neither very low nor very high. To some extent it is consistent with the spirit of the results of some papers in that higher targets do not necessarily signal a higher quality. For example, Mollick (2014) and Cordova and Dolci (2015) find that setting higher thresholds does not lead to higher campaign success rates.

Another line of literature looks at the aspects of funders behaviour in terms of monitoring and collecting information about the campaign. Burtch et al. (2013) suggest that crowdfunding participants pay attention to information and information plays a strategic role in crowdfunding. Agrawal et al. (2015) analyze what types of funders pay more attention to previous information in crowdfunding.

In this paper the analyze the role of imperfect information in a competitive environment with crowdfunding, which is a new direction in this line of research.

Finally note that when talking about crowdfunding the media and internet usually provide examples of firms that use it for funding extremely innovative and often very sophisticated Apple-esque<sup>11</sup> products (see, for example, Kumar et al. (2015)). Examples include 3D-printers, electric cars, smart watches etc. In this case firms retain monopoly power during the development stage and sale of the project (see Santos (2017) among others). Less attention is paid to companies that use crowdfunding for financing more traditional products and services such as handbags,

<sup>&</sup>lt;sup>11</sup> https://artofthekickstart.com/kickstarter-vs-indiegogo-and-how-to-decide-for-your-crowdfunding-campaign/

perfumes, toys etc.<sup>12</sup> In this case the level of competition with producers of similar products increases. Theoretical research on crowdfunding has generally followed this trend and has mostly assumed that crowdfunding is used under monopoly conditions (see, for example, Belleflamme et al. (2014); Chang (2016); Strausz (2017); Chemla and Tinn (2019) or Miglo and Miglo (2019)). However, much less is known about the role of crowdfunding in explaining the behaviour of entrepreneurs operating in competitive markets including cases when entrepreneurs who use crowdfunding compete against entrepreneurs who do not. In this article we compare the role and the importance of crowdfunding in a competitive environment and in the monopoly case.

# 3. Crowdfunding and Monopoly

We begin by considering a traditional framework where an entrepreneurial firm has monopoly power over its product or service. The production is q (all variables are described in Table 1).

| Variable              | Description  |
|-----------------------|--|
| а                     | parameter in the demand function   |
| a <sub>i</sub>        | parameter in the demand function for the level of demand $i, i = h, l$<br>in a model with demand uncertainty |
| μ                     | probability that demand is high  |
| С                     | crowdfunding pre-sales   |
| c <sub>j</sub>        | crowdfunding pre-sales for firm $j, j = 1, 2$ in the model with two firms                                    |
| β                     | benefits for backers   |
| S                     | spot sales   |
| s <sub>j</sub>        | spot sales for firm $j, j = 1, 2$ in the model with two firms  |
| 9                     | quantity produced, $q = c + s$   |
| <i>q</i> <sub>j</sub> | quantity produced for firm $j, j = 1, 2$ in the model with two firms   |
| р                     | spot price   |
| <i>p</i> <sub>c</sub> | crowdfunding price   |
| f                     | platform fee per unit sold   |
| F                     | platform up-front fee  |
| П                     | firm profit  |
| $\Pi_j$               | firm <i>j</i> profit in the model with two firms, $j = 1, 2$   |
| С                     | maximal number of backers in crowdfunding stage  |
| В                     | the cost of crowdfunding campaign failure  |
| π                     | probability of campaign success  |

Table 1. Notations description.

The firm trades on the spot market (the price is p) and (prior to that) it can use a crowdfunding campaign (the crowdfunding or pre-sale price is denoted by  $p_c$ ). Let c and s denote crowdfunding pre-sales and spot sales respectively: q = c + s. If a firm uses crowdfunding, the funders (those who pre-order the product during the pre-sale/crowdfunding stage) expect to receive an extra-benefit

<sup>&</sup>lt;sup>12</sup> https://www.indiegogo.com/projects/public-goods-revolutionizing-household-products--3#/; https://www.indiegogo.com/projects/kantala-handbags-inspired-by-traditional-artisans-vegan#/

(reward)  $\beta$  from the firm.<sup>13</sup> so the total cost of these benefits for the firm equals  $\beta c$ . Everybody is risk-neutral and the risk-free interest rate equals zero.

The sequence of events is as follows.

- 1. Firm chooses  $c. p_c$  is determined.
- 2. Firm chooses *s*. *p* is determined.

Price determination is driven by the following rules: (1) p = a - c - s; parameter *a* reflects the magnitude of the demand and a higher price leads to a lower demand; (2)  $p_c = Ep$ , where Ep represents the market's expectation regarding the spot sale price (non-arbitrage condition). We assume that  $\beta$  is just large enough to compensate funders for the waiting time between the pre-sale stage and the actual sale of the product so the non-arbitrage condition holds.

An equilibrium is a situation where the following hold: (1) p = Ep (rational expectations); (2) during the spot sale period the firm selects *s* to maximize *ps* (spot sale incentive constraint);<sup>14</sup> (3) during the crowdfunding campaign the firm selects *c* to maximize its total profit over two periods  $p_cc + EpEs - \beta c$  (crowdfunding incentive constraint). Here *Es* represents the firm's expectation of *s*. In equilibrium we should have Es = s.

We start the solution by working backwards. When selecting *s*, the firm maximizes ps = (a - c - s)s.

The solution is:

$$s = \frac{a-c}{2} \tag{1}$$

Also

$$p = a - c - s = \frac{a - c}{2} \tag{2}$$

When selecting *c*, the firm maximizes  $p_c c + EpEs - \beta c$ . The market participants can rationally anticipate the spot price and sales using (1) and (2). This implies that when selecting *c*, the firm maximizes  $\frac{a-c}{2}(c + \frac{a-c}{2}) - \beta c$ . The solution is

c = 0

Lemma 1. Under a monopoly, crowdfunding is not used.

Existing monopoly literature on crowdfunding incorporates additonal crowdfunding features to explain its usage. Some examples include non-monetary benefits as in Belleflamme et al. (2014), different market imperfections as in Miglo and Miglo (2019), moral hazard issues as in Strausz (2017) and demand uncertainty as in Chemla and Tinn (2019) etc.

In Belleflamme et al. (2014), a monopoly is facing individual customers with different demand functions. A potential consumer's surplus from buying the product is v - p, where p is the price and v is the consumer's product valuation. Each consumer only needs one unit of the product/service. The valuation from consuming an extra-unit is zero. Consumers buy/order the product/service as long as they have a non-negative surplus v - p, where p is the price. v is uniformly distributed between 0 and a. In this setting if the price equals p, all consumers with v greater than p will buy the product making the demand q = a - p like in our model. Further, on the spot market, the remaining customers (who did not order during crowdfunding) will buy if  $v > p_s$ , where  $p_s$  is the spot price. If  $p_s \ge p$ , no customers will buy on the spot market because they already ordered during crowdfunding, so

<sup>&</sup>lt;sup>13</sup> It may include discounts on the firm's products/services, early access to some of its services, exclusive access to some services etc. Also note that as was mentioned previosuly there exists two types of reward-based crowdfunding: AON and KIA. We focus on KIA. In Section 7 we provide some comments about how the difference between AON and KIA can affect the model's results,

<sup>&</sup>lt;sup>14</sup> For simplicity, production costs are not considered. In Section 7 and Appendix A.4, we argue that it does not affect the model's results.

the crowdfunding model becomes essentially just a spot market model (i.e., there is no special role for crowdfunding: similar to our result in Lemma 1). The only case that makes sense is the case  $p_s < p$ . However this case contradicts the no-arbitrage condition for the equilibrium concept: why would consumers buy for a higher price during crowdfunding and not wait until the spot sales begin? To overcome this problem the authors introduce non-monetary benefits for funders participating in crowdfunding. Although some researchers support the significance of these benefits (see, for example, Schwartz (2015)), others find that their role seems to be negligible (see, for example, Cholakova and Clarysse (2015)).

# 4. A Market with Competition

Suppose that there are two firms each producing one product/service. The production of Firm 1 is  $q_1$  and that of Firm 2 is  $q_2$ . Let  $c_i$  and  $s_i$  denote crowdfunding pre-sales and spot sales respectively for Firm  $i, i \in 1, 2$ :  $q_i = c_i + s_i$ . The sequence of events in the game is as follows.

- 1. Firms decide whether to use crowdfunding (this strategy will be denoted CF) or not (S).
- 2. Firms observe each other's decisions.
- 3. Firms choose  $c_i$ .  $p_c$  is determined.
- 4. Firms choose  $s_i$ . p is determined.

Price determination is driven by the following rules: (1)  $p = a - s_1 - s_2 - c_1 - c_2$ ; (2)  $p_c = Ep$ , where Ep represents the market participants' expectations regarding the spot sale price.

An equilibrium is a situation where the following hold: (1) p = Ep; (2) during the spot sale Firm i, i = 1, 2 selects  $s_i$  to maximize  $ps_i$  for a given choice of  $s_j$  by Firm j,  $j \neq i$ ; (3) during the crowdfunding campaign Firm i, i = 1, 2 selects  $c_i$  to maximize its total profit over two periods  $\Pi_i = p_c c_i + EpEs_i - \beta c_i$  for a given choice of  $c_j$  by Firm j,  $j \neq i$ . Here  $Es_i$  is a firm's expectation of  $s_i$ . In equilibrium we should have  $s_i = Es_i$ ; (4) In stage 1 each firm selects strategy S or CF which maximizes its payoff given the choice of the other firm.

Note that in addition to the non-arbitrage and rational expectation conditions, which are similar to the ones that we used in the monopoly analysis, here we also use a Nash equilibrium concept (similar to traditional literature on duopoly) to analyze the outcomes of the different stages of competition.

In stage 1 the following situations can occur: both firms select S; both firms select CF; Firm 1 selects CF and Firm 2 selects S; Firm 2 selects CF and Firm 1 selects S. If both firms select S, the equilibrium outcome is  $s_1 = s_2 = \frac{a}{3}$  and  $\Pi_1 = \Pi_2 = \frac{a^2}{9}$ . Indeed Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - s_2)$ , which makes:

 $s_i = \frac{a}{3}, i = 1, 2$ 

$$s_1 = \frac{a - s_2}{2}$$
 (3)

Similarly, for Firm 2 we get

 $s_2 = \frac{a - s_1}{2} \tag{4}$ 

Solving (3) and (4) produces

Also

$$p = a - q_1 - q_2 = \frac{a}{3}$$
$$\Pi_i = p_i q_i = \frac{a^2}{9}, i = 1, 2$$
(5)

The firm's profit (and supply) increases with *a* (the intercept of the demand function).

Now consider a situation where both firms select CF. We begin the solution by working backwards. On the spot market, Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - c_1 - s_2 - c_2)$ , which makes:

$$s_1 = \frac{a - c_1 - s_2 - c_2}{2} \tag{6}$$

J. Risk Financial Manag. 2020, 13, 39

Similarly for Firm 2 we get

$$s_2 = \frac{a - c_2 - s_1 - c_1}{2} \tag{7}$$

Solving (6) and (7) produces

$$s_i = \frac{a - c_1 - c_2}{3}, i = 1, 2 \tag{8}$$

During crowdfunding Firm 1 maximizes  $(s_1 + c_1)(a - s_1 - c_1 - s_2 - c_2) - \beta c_1$  subject (8). This gives us:

$$c_1 = \frac{a - c_2 - 9\beta}{4}$$
$$c_2 = \frac{a - c_1 - 9\beta}{4}$$

This implies<sup>15</sup>:

Similarly for Firm 2:

$$c_i = \frac{a - 9\beta}{5}, i = 1, 2 \tag{9}$$

(8) and (9) imply that the prices equal

$$p_i = a - s_1 - c_1 - s_2 - c_2 = \frac{a + 6\beta}{5}, i = 1, 2$$

$$\Pi_i = p_i q_i - \beta c_i = \frac{2a^2 + 4a\beta + 9\beta^2}{25}, i = 1, 2$$

Now consider a situation where one firm (for instance, Firm 1) selects CF and the other one selects S. In this case  $c_2 = 0$ .

Since  $c_2 = 0$ , (8) implies:

$$s_i = \frac{a - c_1}{3}, i = 1, 2 \tag{10}$$

During crowdfunding Firm 1 maximizes  $(s_1 + c_1)(a - s_1 - c_1 - s_2) - \beta c_1$  subject (10). This gives us:

$$c_1 = \frac{a - 9\beta}{4} \tag{11}$$

Accordingly we have

$$\Pi_1 = \frac{a^2 - 2a\beta + 9\beta^2}{8}$$
$$\Pi_2 = s_2 p = \frac{a^2 + 6a\beta + 9\beta^2}{16}$$

**Proposition 1.** (1) If  $\beta \geq \frac{a}{9}$ . the equilibrium is (S,S); (2) Otherwise the equilibrium is (CF,CF).

**Proof.** Consider the matrix of payoffs.

The case  $\beta \ge \frac{a}{9}$  was discussed in footnote 15. Consider  $\beta < \frac{a}{9}$ . It follows from Figure 1 that switching to crowdfunding from the case (S,S) improves the payoff of Firm 1. Early committeent of Firm 1 makes the product price lower, increases the production of Firm 1 and forces Firm 2 out of its

<sup>&</sup>lt;sup>15</sup> Note that (9) implies that  $\beta$  should be relatively low (i.e.,  $a > 9\beta$ ). If it is not, a corner ("uninteresting") solution arises where crowdfunding is not used at all. Throuought the article we mostly assume that this condition holds.

optimal quantity. So (S,S) is not an equilibrium. Indeed the profit of Firm 1 in the case (S,CF) is greater than it is in the case (S,S) if:

$$\frac{a^2-2a\beta+9\beta^2}{8}>\frac{a^2}{9}$$

This holds because it can be written as

$$\frac{(a-9\beta)^2}{72} > 0$$

| Firm 1/Firm 2 | S   | CF   |
|---------------|---|--|
| S             | $\frac{a^2}{9}; \frac{a^2}{9}$                                | $\left  \frac{(a+3\beta)^2}{16}; \frac{a^2-2a\beta+9\beta^2}{8} \right $       |
| CF            | $\frac{a^2 - 2a\beta + 9\beta^2}{8}; \frac{(a+3\beta)^2}{16}$ | $\frac{2a^2 + 4a\beta + 27\beta^2}{25}; \frac{2a^2 + 4a\beta + 27\beta^2}{25}$ |

#### Figure 1. Matrix of payoffs.

As follows from the matrix of payoffs, whether the equilibrium is (S,CF) or (CF,CF) depends on the following:

$$\frac{2a^2 + 4a\beta + 27\beta^2}{25} > \frac{(a+3\beta)^2}{16} \tag{12}$$

If this holds, the equilibrium is (CF,CF) and vice versa. (12) can be written as

$$\frac{(a-9\beta)(7a-23\beta)}{400} > 0$$

which holds because  $a > 9\beta$  so (CF,CF) is an equilibrium.  $\Box$ 

This section demonstrates that crowdfunding will be used in equilibrium. A situation where both competitors do not use crowdfunding, and only use the spot market, is not an equilibrium because each firm has an incentive to deviate and use a crowdfunding campaign prior to its spot sales. Early commitment of a firm increases its production, makes the product price lower and forces the other firm out of its optimal quantity. As a result, firms fight agressively for market share by using crowdfunding. This is a new result because it argues that in an environment without any non-monetary benefits (as in Belleflamme et al. (2014)), moral hazard issues (as in Strausz (2017) or Chemla and Tinn (2019)) or other imperfections (as in Miglo and Miglo (2019)), crowdfunding can still provide value for firms in a competitive environment. This contrasts the monopoly case discussed in the previous section. As was mentioned previously crowdfunding has never been introduced/analyzed in a competitive model so this is a new result. We provide more discussion in Section 7.

The next step in the analysis is based on the following observations. First, for some values of  $\beta$ , crowdfunding is not used in equilibrium. Second, in many cases equilibria with crowdfunding are not Pareto-efficient.<sup>16</sup> Because of that and also because our model is the first to analyze the role of crowdfunding in a competitive environment, it is interesting and important to analyze other aspects of crowdfunding in shaping the outcome of competition. So far it has been assumed that information is complete and perfect. In the next section we will analyze the model with imperfect information and demonstrate some interesting results related to the use of crowdfunding.

<sup>&</sup>lt;sup>16</sup> It follows from Figure 1 that the equilibrium may be Pareto-inefficient. Indeed Pareto-efficiency of (CF,CF) depends on the following:  $\frac{2a^2+4a\beta+27\beta^2}{25} > \frac{a^2-2a\beta+9\beta^2}{8}$ . If this holds, (CF,CF) is Pareto-efficient and vice versa. This can be written as  $\frac{(a-9\beta)(\beta-9a)}{200} > 0$ . This does not hold because  $a > 9\beta$ .

#### 5. Imperfect Information

Suppose that in our model firms face demand uncertainty:  $a = a_h$  with probability  $\mu$  and otherwise  $a = a_l, a_h > a_l$ . Two cases are possible. One where the demand is unknown to both competitors and one where one of competitors receives private information about demand (without loss of generally let this be Firm 1). We begin by considering the latter. Firm 1 knows the demand and respectively Firm 1 can be two types depending on its knowledge:  $a = a_h$  for type h and  $a = a_l$  for type l. Firm 1 knows the value of a while Firm 2 does not (although Firm 2 believes that Firm 1 is type h with probability  $\mu$ ). The analysis of this scenario is interesting and useful for at least two reasons. The uninformed party receives information through two channels: one is related to its own effort (e.g., conducting a crowdfunding campaign and then analyzing the demand) and the second channel is observing the actions of the informed party (Firm 1). Our analysis will help to understand the interaction of both channels and their impact on the outcome of competition. Secondly it is interesting to analyze the actions of the different types of the informed player (Firm 1) because it can generate predictions about the connections between the firm's strategy and its qualities. It is an important part of crowdfunding research as well as other literature dealing with different ways of rasing funds by firms.

An equilibrium is a situation where no type has an incentive to deviate. It is characterized by the actions (strategies) undertaken by each type of Firm 1, the beliefs of Firm 2 about Firm 1's type after observing different actions (on equilibrium path and off-equilibrium path) and the actions undertaken by Firm 2. We first focus on separating equilibria which help to generate predictions about the signalling power of crowdfunding. It is an equilibrium where the different types of Firm 1 select different strategies. Two possible separating equilibria may exist. One where type *l* uses crowdfunding and type *h* does not and another where type *h* uses it and *l* does not.

**Proposition 2.** (1) A separating equilibrium where type *l* selects CF and type *h* selects *S* does not exist; (2) if and only if  $\beta$  is sufficiently small, a separating equilibrium exists where type *h* selects CF and type *l* selects *S*.

#### **Proof.** See Appendix A.1. $\Box$

To illustrate Proposition 2, consider the case  $\beta = 0$ . The matrix of payoffs for the case of perfect information for each type of Firm 1 is as follows.

The candidate for a separating equilibrium is the case where Firm 2 selects CF and the different types of Firm 1 select different strategies. The situation where Firm 2 selects S cannot be an equilibrium because looking at the matrix of payoffs we can see that Firm 2 would deviate to CF. One can show that a separating equilibrium where type h selects CF and type l selects S exists and a separating equilibrium where type l selects CF and type h selects S does not exist. Consider a situation where Firm 1 selects CF only if it is type l and selects S otherwise. Firm 2 believes that the Firm 1's type is l when observing CF and h when observing S. From Figure 2 the profits of each type are

$$\Pi_{1l} = \frac{2a_l^2}{25} \tag{13}$$

$$\Pi_{1h} = \frac{a_h^2}{16} \tag{14}$$

| Firm 1 (type <i>l</i> )/Firm 2 | S   | CF   |
|--------------------------------|---|--|
| S                              | $\frac{a_l^2}{9}; \frac{a_l^2}{9}$  | $\frac{a_l^2}{16}; \frac{a_l^2}{8}$                          |
| CF                             | $\frac{a_l^2}{8}; \frac{a_l^2}{16}$   | $\frac{2}{5} \mid \frac{2a_l^2}{25}; \frac{2a_l^2}{25} \mid$ |
|                                |   |  |
|                                |   |  |
| Firm 1 (type <i>h</i> )/Firm 2 | S   | CF   |
| Firm 1 (type <i>h</i> )/Firm 2 | $\begin{array}{  } S \\ \hline a_h^2 \\ \hline 9; 9 \\ \hline \end{array}$          | CF   |
| Firm 1 (type <i>h</i> )/Firm 2 | $\begin{array}{  } S \\ \hline a_h^2 \\ \hline 9 \\ \hline 2 \\ \hline \end{array}$ | $CF$ $\frac{a_h^2}{16}; \frac{a_h^2}{8}$                     |

Figure 2. Matrix of payoffs for different types of firm.

Suppose that h mimics l and uses crowdfunding. (9) implies

$$c_2 = \frac{a_l}{5}$$

After learning new information about demand from observing the results of the crowdfunding stage Firm 2 realizes that its rival is type *h* (i.e., the demand is high), and so it chooses  $s_2$  to maximize  $s_2(a_h - s_1 - s_2 - c_1 - c_2)$ , which makes:

$$s_2 = \frac{5a_h - 5s_1 - 5c_1 - a_l}{10} \tag{15}$$

Similarly for Firm 1 we get

$$s_1 = \frac{5a_h - 5s_2 - 5c_1 - a_l}{10} \tag{16}$$

Solving (15) and (16) produces

$$s_i = \frac{a_h - c_1 - \frac{a_l}{5}}{3} = \frac{5a_h - 5c_1 - a_l}{15}$$

Also

$$p = a_h - 2\frac{5a_h - 5c_1 - a_l}{15} - c_1 - \frac{a_l}{5} = \frac{5a_h - 5c_1 - a_l}{15}$$

During crowdfunding Firm 1 maximizes  $\left(\frac{5a_h-5c_1-a_l}{15}+c_1\right)\frac{5a_h-5c_1-a_l}{15}$ . This gives us:

$$c_1 = \frac{5a_h - a_l}{20}$$
$$p = \frac{5a_h - a_l}{20}$$

Firm 1's profit then equals

$$\Pi_{1h} = \left(\frac{5a_h - a_l}{20} + \frac{5a_h - a_l}{20}\right)\frac{5a_h - a_l}{20} = \frac{(5a_h - a_l)^2}{200}$$
(17)

Comparing this with (14) we find that it is smaller if

$$\frac{2a_l}{5(2-\sqrt{2})} > a_h \tag{18}$$

This does not hold since  $\frac{2a_l}{5(2-\sqrt{2})}$  is smaller than  $a_l$  and we know  $a_h \ge a_l$ . So this equilibrium does not exist. If Firm 2 perceives a firm that uses a crowdfunding campaign to be a low demand firm, it will be pessimistic about the price and will concede the market. Type *h* can benefit from this situation by mimicking the strategy of type *l* during the crowdfunding stage.

J. Risk Financial Manag. 2020, 13, 39

Now consider an equilibrium where type h uses crowdfunding and l does not. The equilibrium payoffs are:

$$\Pi_{1h} = \frac{2a_h^2}{25} \tag{19}$$

$$\Pi_{1l} = \frac{a_l^2}{16}$$
(20)

Suppose that *h* mimics *l* and does not use crowdfunding. Similarly to (11) we get<sup>17</sup>:

$$c_2 = \frac{a_l}{4}$$

After learning information about demand, Firm 2 chooses  $s_2$  to maximize  $s_2(a_h - s_1 - s_2 - c_2)$ , which makes:

$$s_2 = \frac{a_h - s_1 - \frac{u_l}{4}}{2} \tag{21}$$

Similarly for Firm 1 we get

 $s_1 = \frac{a_h - s_2 - \frac{a_l}{4}}{2} \tag{22}$ 

Solving (21) and (22) produces

 $s_i = \frac{4a_h - a_l}{12}$ 

Also

$$p = a_h - 2\frac{4a_h - a_l}{12} - \frac{a_l}{4} = \frac{4a_h - a_l}{12}$$
$$\Pi_{1h} = \frac{(4a_h - a_l)^2}{144}$$
(23)

Comparing this with (19) we find that it is smaller if

$$a_h < \frac{5}{4(5 - 3\sqrt{2})} a_l \tag{24}$$

Suppose that l mimics h and uses crowdfunding. The calculations are similar to the previously analyzed equilibrium case where type h plays CF and type l mimics h. Therefore (17) implies

$$\Pi_{1l} = \frac{(5a_l - a_h)^2}{200}$$

Comparing this with (20) we find that it is smaller if

$$a_h > \frac{5(2-\sqrt{2})}{2}a_l \tag{25}$$

Note that conditions (24) and (25) can hold simultaneously because  $\frac{5(2-\sqrt{2})}{2} < \frac{5}{4(5-3\sqrt{2})}$ . Therefore an equilibrium where type *h* uses crowdfunding and type *l* does not may exist.

The main result of Proposition 2 is that the only signalling equilibrium that can exist is the one where the high-demand type selects crowdfunding (and the low-demand type selects spot sales).

Next we analyze the pooling equilibria. We define a pooling equilbrium as one where both types of Firm 1 select the same strategy. We will also check that the off-equilibrium beliefs of Firm 2 survive the intuitive criterion of Cho and Kreps (1987). This condition means that Firm 2's off-equilibrium

<sup>&</sup>lt;sup>17</sup> (11) applies to the case where Firm 1 plays CF and Firm 2 plays S. Here it is an opposite case.

path beliefs are reasonable in the sense that if for any Firm 1 type its maximal payoff from deviation is not greater than its equilibrium payoff then Firm 2 should place a probability of 0 on this firm making a deviation. The definitions above are consistent with the standard perfect bayesian equilibrium definition (see, for instance, Fudenberg and Tirole (1991)) with the addition of the intuitive criterion, which is quite common in these types of games involving financing, i.e., when firms raise funds from external investors (see, for instance, Nachman and Noe (1994)). If multiple equilibria exist we will use the mispricing criterion to indicate the one that is most likely to exist. We use the standard concept of mispricing that can be found, for example, in Nachman and Noe (1994). The magnitude of mispricing (usually it is the difference between a firm's value under symmetric information and its equilibrium value) in a given equilibrium is equal to that of the undervalued type(s). The mispricing of the overvalued type(s) does not matter.

**Proposition 3.** (1) A pooling equilibrium where both types of Firm 1 and Firm 2 use crowdfunding may exist; (2) If it does, it is dominated, according to the mispricing criterion, by the separating equilibrium where h selects CF if and only if  $\mu$  is sufficiently high.

#### **Proof.** See Appendix A.2. $\Box$

The following explains some ideas behind Proposition 3. First a pooling equilibrium with crowdfunding does not always exist. If it does though it should be compared with the separating equilbrium using the mispricing criterion. First note that compared to the symmetric information case, the low-demand firm (type *l*) loses under the pooling equilibrium because its payoff is less than it is under symmetric information. This is because in the case of pooling, Firm 2 is more aggressive compared to the symmetric information case for type *l*, since Firm 2 has higher expectations about product demand. So type *l*'s profit is reduced. To compare the two equilibria using the mispricing criterion we will compare the payoffs of type *l*, which is equal to the firm's value for its founders, in each equilibria. To illustrate this consider the case  $\beta = 0$ . The outcome of *l* under the separating equilibrium (see (20)) equals  $\frac{a_l^2}{16}$ . As shown in Appendix A.2, under pooling its payoff is

$$\frac{((4+\mu)a_l - \mu a_h)^2}{200} \tag{26}$$

It can be seen that the firms' payoff under the pooling equilibrium decreases with  $\mu$  because if Firm 2 perceives Firm 1 to be a high-demand firm with a higher probability it increases its volume of crowdfunding which ultimately reduces the total outcome.<sup>18</sup> This leads to the following condition. The payoff under the separating equilibrium is higher for type *l* than it is under the pooling equilibrium if

$$\mu > \frac{(8 - 5\sqrt{2})a_l}{2(a_h - a_l)}$$

This implies that the separating equilibrium prevails or that signalling opportunities for high-demand firms are more relevant when the economy is performing well ( $\mu$  is high). As a result, our model suggests that the usage of crowdfunding by entrepreneurial firms for signalling is procyclical.

Finally consider the case when demand is unknown by both competitors. We find that in this case the outcome is similar to Proposition 1 except that the likelihood of equilibrium with crowdfunding increases because it adds value to firms by revealing information about the demand during the crowdfunding stage. The following proposition considers the case when neither firm knows the

<sup>&</sup>lt;sup>18</sup> In order for this equilibrium to exist the following should hold:  $(4 + \mu)a_l > \mu a_h$  (see Appendix A.2). This implies that (26) decreases in  $\mu$  because it's derivative with respect to  $\mu$  equals  $a_l - a_h < 0$ .

demand, i.e., the value of *a*. As before we assume that  $a = a_h$  with probability  $\mu$  and otherwise  $a = a_l$ , where  $a_h > a_l$ .

**Proposition 4.** (1) If  $\beta$  is sufficiently small, the equilibrium is (CF,CF); (2) If not, the equilibrium is (CF,CF) if the difference between  $a_h$  and  $a_l$  is large enough. Otherwise, the equilibrium is (S,S).

## **Proof.** See Appendix A.3. $\Box$

Proposition 4 confirms the results in Proposition 1 to some extent. Crowdfunding is used when  $\beta$  is sufficiently low. The last part has an interesting interpretation: a large degree of uncertainty about the demand makes crowdfunding more likely to appear in equilibrium. The result is intuitive. If uncertainty about demand is high, using crowdfunding prior to spot sales helps the firm learn demand and hence helps it in making production decisions later.

#### 6. The Main Implications of The Model and Its Contribution to The Literature

Our paper has several implications for a firm's choice of crowdfunding.

Proposition 1 explains the value of crowdfunding for firms. Under monopoly (and without any additional assumptions such as non-monetary benefits, market imperfections, moral hazard problems etc.), crowdfunding is not used. If the non-argbitrage condition holds, the firm should compensate its buyers for waiting by offering rewards, which makes crowdfunding a more expensive option. However in the case of a duopoly firms may select crowdfunding in favor of just spot price sales. Except some extreme cases when  $\beta$  is large and we have a corner solution where crowdfunding is not a feasible option, a situation where both firms do not use crowdfunding is never an equilibrium. Firms have an incentive to use crowdfunding and consequently force the second firm out of the market to some extent. Early commitment from one firm (i.e., using crowdfunding) makes the product price lower, increases its production and forces the other firm out of its optimal quantity. Furthermore Proposition 4 implies that the likelihood of using crowdfunding increases with demand uncertainty. This is intuitive. A higher level of uncertainty increases the value for the firm to engage with customers early which can have a positive effect on future production and spot sales decisions.

Proposition 2 implies that high-demand firms can use crowdfunding to signal their quality. If an uninformed firm perceives a firm that uses a crowdfunding campaign to be a low-demand firm, it will be pessimistic about the price and will concede the market. A high-demand firm can benefit from this situation by mimicking the strategy of the low-demand firm during the crowdfunding stage. So an equilibrium where the low-demand firm uses crowdfunding and the high-demand firm does not does not exist. However, we show that the opposite is true and an equilibrium where a high-demand firm uses crowdfunding can exist. The uninformed firm (Firm 2) perceives crowdfunding to be a high-demand product with high prices so Firm 2 also increases its production, which may reduce prices implying that a low-demand firm will lose profits if it mimics this strategy. This prediction has not been directly tested but is consistent with the spirit of the results found in Ahlers et al. (2015); Cumming et al. (2019) and Mollick (2014) (the firm's financing choice can serve as a signal of a project's quality).

Proposition 3 implies that the separating equilibrium prevails or that signalling opportunities for high-demand firms are more relevant when the economy is performing well ( $\mu$  is high). As a result, our model suggests that the usage of crowdfunding by entrepreneurial firms for signalling is procyclical. To the best of our knowledge, no empirical research exists testing this prediction directly. Further research is expected. Rampini (2004) is probably closest to the spirit of this prediction since it suggests that entrepreneurial activity is procyclical (see Llopis et al. (2015) for a review of related literature). Also note that, Dushnitsky et al. (2016), for example, find that crowdfunding development is closely related to the level of entrepreneurial activities.

Crowdfunding can also be viewed as an example of a firm's capital structure decision. In light of capital structure literature (see, for example, Harris and Raviv (1991) for a review) our paper falls into

the branch where capital structure decisions are connected with production decisions in a competitive setting. Brander and Lewis (1986); Bolton and Scharfstein (1990) and Williams (1995) consider the role of capital structure in improving a firm's competitive position and find that the level of a firm's debt is strongly related to its level of agressiveness on the product market. Our paper makes two contributions. First it introduces reward-based crowdfunding as a financing choice for a firm. Existing literature usually focuses on the traditional debt/equity choice. Second, to the best of knowledge this literature does not analyze the role of asymmetric information, which is a major part of our model.

#### 7. Model Extensions and Robustness

#### 7.1. Different Types of Market Structure

This article has used a Cournot style competiton where firms chose the quantities of their products and not their prices. A model with a Bertrand style competition could also be considered. Our analysis shows that although quantitatively there may be differences in some propositions overall the main results will still hold. Also it seems like literature usually suggests using Bertrand when firms have a lot of production flexibility (see, for example, Dixon (2001)) but this does not seem to be the case for crowdfunding firms which, in most cases, experience delivery delays (Schiavone (2017)).

In terms of our approach to modelling the demand side note that our focus in this article is to analyze the role of crowdfunding in a competitive environment. That is why we adopt a relatively simple demand function. Alternatively, a significantly different approach to model the demand side can be taken where individual customers with different demand functions are included (see, for example, Belleflamme et al. (2014) or Strausz (2017)). This approach is often used in industrial organization and price discrimination literature. Note, however, that as we mentioned previously, a simple introduction of individual customers often leads to the same results (see a discussion in Section 3). Below we analyze an example with individual demands and show that the basic idea holds.

An entrepreneurial firm is considering selling its product to potential customers. A potential consumer's surplus from buying the product is v - p, where p is the price and v is the consumer's product valuation. Each consumer only needs one unit of the product/service. The valuation from consuming an extra-unit is zero. Consumers buy/order the product/service as long as they have a non-negative surplus v - p, where p is the price. v is uniformly distributed between 0 and a. As previously: the firm trades on the spot market and (prior to that) it can use a crowdfunding campaign; the pre-sale (crowdfunding) price equals p; and if a firm uses crowdfunding, the funders expect to receive a reward  $\beta$  from the firm,  $\beta \leq a$ . Also we assume that if a consumer is indifferent between buying during crowdfunding and spot selling, they will split randomly between periods.

Suppose the price is *p*. All buyers with  $v \ge p$  are interested in buying the product. By assumption 50% of them will be buying during crowdfunding and 50% during pre-sale. Since they are randomly split, the firm's profit equals:  $p\frac{a-p}{2} - \beta\frac{a-p}{2} + p\frac{a-p}{2} = p(a-p) - \beta\frac{a-p}{2}$ . The optimal price is  $p = \frac{a+\beta/2}{2}$ . The firm's profit is

$$\frac{a+\beta/2}{2}\left(a-\frac{a+\beta/2}{2}\right)-\beta\frac{a-\beta/2}{4}=\frac{a^2}{4}+\frac{\beta^2}{16}-\frac{a\beta}{4}$$
(27)

If the firm does not use crowdfunding all consumers with  $v \ge p$  will buy the product on the spot market. The firm's profit equals p(a - p). The optimal price is  $p = \frac{a}{2}$ . The firm's profit is  $\frac{a}{2}(a - \frac{a}{2}) = \frac{a^2}{4}$ . This is greater than (27). So the firm should not use crowdfunding which is consistent with Lemma 1.

Now consider a duopoly. We assume that if there is more than one firm offering the same price, consumers will be split randomly. Suppose that both firms do not use crowdfunding. By assumption 50% of them will be from Firm 1 and 50% from Firm 2. Since they are randomly split, each firm's profit equals  $p\frac{a-p}{2}$ . The optimal price is  $p = \frac{a}{2}$ . The firm's profit is

$$\frac{a}{2}\left(\frac{a-\frac{a}{2}}{2}\right) = \frac{a^2}{8} \tag{28}$$

Now suppose that one firm deviates and uses crowdfunding (for example, Firm 1). By assumption 50% of consumers will buy the product during crowdfunding and 50% during the spot market sale. Firm 1's profit equals:  $\frac{a}{2}\frac{3a}{8} - \beta \frac{a}{4} = \frac{3}{16}a^2 - \beta \frac{a}{4}$ . This greater than (28). So an equilibrium where both firms do not use crowdfunding does not exist.

Now consider the case where only one firm uses crowdfunding (for example, Firm 1). By assumption 50% of consumers will be buying the product during crowdfunding and 50% during the spot market sale. Firm 1's profit equals:  $p\frac{3(a-p)}{4} - \beta\frac{a-p}{2}$ . The optimal price is  $p = \frac{\frac{3a}{4} + \frac{\beta}{2}}{\frac{3}{2}}$ . The firm's profit is  $\frac{a+\beta/2}{2}(a - \frac{a+\beta/2}{2}) - \beta\frac{a-\beta/2}{4}$ . The other firm's profit is  $\frac{\frac{3a}{4} + \frac{\beta}{2}}{\frac{3}{2}}\frac{(a-p)}{4}$ .

Finally consider the case where both firms use crowdfunding. By assumption 50% of consumers will buy the product during crowdfunding and 50% during the spot market sale. The firm's profit equals:  $p\frac{a-p}{4} - \beta\frac{a-p}{4} + p\frac{a-p}{4} = \frac{p(a-p)}{2} - \beta\frac{a-p}{4}$ . The optimal price is  $p = \frac{a+\beta/2}{2}$ . The firm's profit is

$$\frac{a+\beta/2}{2}\left(\frac{a}{2}-\frac{a+\beta/2}{4}\right)-\beta\frac{a-\beta/2}{8}=\frac{a^2}{8}+\frac{\beta^2}{32}-\frac{a\beta}{8}$$
(29)

Suppose that one firm deviates and does not use crowdfunding (for example, Firm 1). By assumption 50% of consumers will buy the product during crowdfunding and 50% during the spot market sale. Firm 1's profit equals:  $\frac{a+\beta/2}{2}(a-\frac{a+\beta/2}{2})/4 = \frac{a^2}{16} - \frac{\beta^2}{64}$ . This is smaller than (29). So an equilibrium where both firms use crowdfunding does exist, which is consistent with Proposition 1.

#### 7.2. Platform Fees

In this section we analyze the role of crowdfunding platform fees. We assume that the firm profit equals  $\Pi = pq - \beta c - F(c)$ , where F(c) = F + fpc if c > 0 and 0 otherwise; f represents a crowdfunding platform's fee as a percentage of funds raised and F is its fixed up-front fee. Consider the case with symmetric information. For simplicity the case  $\beta = 0$  is considered. Qualitatively the conlcusions do not change when  $\beta > 0$ .

**Proposition 5.** *The matrix of payoffs is:* 

| Firm 1/Firm 2 | S   | CF  |
|---------------|---|---|
| S             | $\frac{a^2}{9}; \frac{a^2}{9}$            | $\left  \frac{a^2(1-f)^2}{(4-6f)^2}; \right $   |
|               |   | $\frac{a^2(2-5f+3f^2)(1-f)}{(4-6f)^2} - F$  |
| CF            | $\frac{a^2(2-5f+3f^2)(1-f)}{(4-6f)^2}-F;$ | $\frac{a^2(6-21f+19f^2+6f^3)(3-6f+13f^2)}{9(5-14f+9f^2)^2}-F;$  |
|               | $\frac{a^2(1-f)^2}{(4-6f)^2}$             | $\left  \begin{array}{c} \frac{a^2(6-21f+19f^2+6f^3)(3-6f+13f^2)}{9(5-14f+9f^2)^2} - F \end{array} \right $ |

Figure 3. Matrix of payoffs with platform fees.

**Proof.** If both firms select S, the equilibrium outcome is  $s_1 = s_2 = a/3$  and  $\Pi_1 = \Pi_2 = a^2/9$  (same as without platform fees).

Now consider a situation where both firms select CF. On the spot market, Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - s_2 - c_1 - c_2)$ , which makes:

$$s_1 = \frac{a - s_2 - c_1 - c_2}{2} \tag{30}$$

Similarly for Firm 2 we get

$$s_2 = \frac{a - s_1 - c_1 - c_2}{2} \tag{31}$$

#### J. Risk Financial Manag. 2020, 13, 39

Solving (30) and (31) produces

$$s_i = \frac{a - c_1 - c_2}{3} \tag{32}$$

During crowdfunding Firm 1 maximizes  $(s_1 + c_1)p - fpc_1 - F = (s_1 + c_1)(a - s_1 - s_2 - c_1 - c_2) - f(a - s_1 - s_2 - c_1 - c_2)c_1 - F = (s_1 + c_1 - fc_1)(a - s_1 - s_2 - c_1 - c_2) - F$  subject (32). This gives us:

$$c_1 = \frac{a - c_2 - 3af + 3fc_2}{4 - 6f}$$

Similarly for Firm 2:

$$c_2 = \frac{a - c_1 - 3af + 3fc_1}{4 - 6f}$$

It implies:

$$c_i = \frac{a - 4af - 2af^2}{5 - 14f + 9f^2} \tag{33}$$

From (32) we get:

$$s_i = \frac{3a - 6af + 13af^2}{3(5 - 14f + 9f^2)}$$

(32) and (33) imply that the price equals

$$p = \frac{3a - 6af + 13af^2}{3(5 - 14f + 9f^2)}$$

The firm's profit equals then

$$\Pi_i = \frac{a^2(6 - 21f + 19f^2 + 6f^3)(3 - 6f + 13f^2)}{9(5 - 14f + 9f^2)^2} - F$$
(34)

Now consider a situation where one firm (for instance, Firm 1) selects CF and the other one selects S. In this case  $c_2 = 0$ .

Since  $c_2 = 0$ , (32) implies:

$$s_1 = s_2 = p = \frac{u - c_1}{3}$$
$$\Pi_1 = (c_1 + \frac{a - c_1}{3} - fc_1)\frac{a - c_1}{3} - F = \frac{(a + 2c_1 - 3fc_1)(a - c_1)}{9} - F$$
(35)

Firm 2 profit equals

$$\Pi_2 = s_2 p = \frac{(a - c_1)^2}{9}$$

Optimal  $c_1$  that maximizes Firm 1 expected profit (given by (35)) equals

$$c_1 = \frac{a(1-3f)}{4-6f} \tag{36}$$

Accordingly we have  $\Pi_2 = \frac{a^2(1-f)^2}{(4-6f)^2}$  and  $\Pi_1 = \frac{a^2(2-5f+3f^2)(1-f)}{(4-6f)^2} - F$ .  $\Box$ 

It follows rather obviously from Figure 3 that if *F* is very large then the CF strategy will not be used. Interestingly, however, if *F* is sufficiently low then regardless of the value of *f* Proposition 1 stands, ie. (S,S) is not an equilibrium. Indeed consider F = 0. If a firm deviates to CF its payoff equals

 $\frac{a^2(2-5f+3f^2)(1-f)}{(4-6f)^2}$  and it is greater than  $\frac{a^2}{9}$  for any value of f.<sup>19</sup> Indeed, the difference between these two values can be written as  $\frac{a^2(1-3f)^2}{36(2-3f)^2} > 0$ . So the result of our model holds (Proposition 1). Also, the optimal amount of crowdfunding in equilibrium depends on f but not F. In some cases a higher f may imply a higher profit.

## Corollary 1. A higher fee may lead to higher firm's profit in equilibrium.

**Proof.** This follows from the analysis of (34). For example when f = 0,  $\Pi = 2/25$  but when f = 0.15,  $\Pi > 0.09 > 2/25$ .  $\Box$ 

This is because the higher f reduces the crowdfunding level which may be beneficial for producers. Firms use crowdfunding in equilibrium but this equilibrium is not Pareto-improving compared to the situation where both firms use spot sales exclusively (Proposition 1). Reducing the optimal amount of crowdfunding means that firms behave less aggressively and more cooperatively ("stealing" less from each other). Interestingly this is consistent with the observation that most crowdfunding platforms do not charge a fixed fee up front but rather a per unit fee or a per value fee.<sup>20</sup> It may also explain why platforms charge relatively high fees for firms that use Marketplace (on Indiegogo) for example.

## 7.3. Fixed Start-Up Costs

Suppose that the fixed costs of launching production equals I > 0. The crowdfunding campaign needs to then cover both fixed and variable costs of production. The analysis of this extension is largely identical to the model analysis with lump sum fee cost F from the previous section. The only interesting conclusion is that if I is large enough (also similar to that in Section 7.2 regarding F) the firm will not be able to use crowdfunding. This is consistent in general with Belleflamme et al. (2014) and Miglo and Miglo (2019) where reward-based crwodfunding can not be used if I is large enough.

#### 7.4. Risk of Crowdfunding Campaign Failure

In Proposition 4 we studied a case where information is symmetric but imperfect, i.e., each firm faced the demand uncertainty. We demonstrated that increasing uncertainty of demand makes the usage of crowdfunding more probable. In this section we extend the analysis of risk. We focus on the risk of crowdfunding campaign. It may not necessarily be related to demand factors but, for example, to inefficient platform choice or campaign mismanagement etc. Let us assume that the campaign succeeds with probability  $\pi$  and respectively fails with probability  $1 - \pi$ . The firm profit is 0 in the latter case. Later we will discuss the case when the failure is costly.

In stage 1 the following situations can occur: both firms select S; both firms select CF; Firm 1 selects CF and Firm 2 selects S; Firm 2 selects CF and Firm 1 selects S. If both firms select S, the equilibrium outcome is  $s_1 = s_2 = \frac{a}{3}$  and  $\Pi_1 = \Pi_2 = \frac{a^2}{9}$ .

Now consider a situation where both firms select CF. The scenario on the spot market depends on the outcome of crowdfunding campaigns. If both campaings succeeded, both firms enter the spot market. Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - c_1 - s_2 - c_2)$ , which makes:

$$s_1 = \frac{a - c_1 - s_2 - c_2}{2} \tag{37}$$

<sup>&</sup>lt;sup>19</sup> As follows from (33) and (36), *f* should be less than  $(\sqrt{6} - 2)/2 \approx 0.21$  in order to avoid a corner solution where crowdfunding is not used at all. In practice, *f* is usually in the range of 3%-15%. See, for example, https://www.thebalancesmb.com/entrepreneurs-guide-to-fees-on-crowdfunding-platforms-985187

<sup>&</sup>lt;sup>20</sup> See, for example, https://support.indiegogo.com/hc/en-us/articles/204456408-Fees-Pricing-for-Campaigners-Howmuch-does-Indiegogo-cost-

J. Risk Financial Manag. 2020, 13, 39

Similarly for Firm 2 we get

$$s_2 = \frac{a - c_2 - s_1 - c_1}{2} \tag{38}$$

Solving (37) and (38) produces

$$s_i = \frac{a - c_1 - c_2}{3}, i = 1, 2 \tag{39}$$

Now consider the scenario when the campaign of one firm succeeded (for example Firm 1) and the other one failed. In this case Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - c_1)$ , which makes:

$$s_1 = \frac{a - c_1}{2}$$
(40)

Taking into account (39) and (40) we find that during crowdfunding Firm 1 maximizes the expected value of its profit  $\Pi_1 = \pi^2((\frac{a+2c_1-c_2}{3})(\frac{a-c_1-c_2}{3}) - \beta c_1) + \pi(1-\pi)((\frac{a+c_1}{2})(\frac{a-c_1}{2}) - \beta c_1)$ . Here  $\pi^2$  is the probability that both campaign succeed and in this case Firm 1 profit equals  $(\frac{a+2c_1-c_2}{3})(\frac{a-c_1-c_2}{3}) - \beta c_1$  and  $\pi(1-\pi)$  is the probability that Firm 1 campaign succeeds and Firm 2 campaign fails in which case Firm 1 profit equals  $(\frac{a+c_1}{2})(\frac{a-c_1}{2}) - \beta c_1$ . This gives us:

$$c_1 = \frac{2\pi(a - c_2) - 18\beta}{9 - \pi}$$

Similarly for Firm 2:

$$c_2 = \frac{2\pi(a - c_1) - 18\beta}{9 - \pi}$$

This implies:

$$c_i = \frac{2a\pi - 18\beta}{9 + \pi}, i = 1, 2 \tag{41}$$

The firms' profits then equal

$$\Pi_{i} = \frac{\pi(3a + a\pi - 6\beta)(27a - 18a\pi - a\pi^{2} + 54\beta - 6\beta\pi)}{4(9 + \pi)^{2}} - \frac{\beta\pi(2a\pi - 18\beta)}{9 + \pi}, i = 1, 2$$
(42)

Now consider a situation where one firm (for instance, Firm 1) selects CF and the other one selects S. In this case  $c_2 = 0$ .

Since  $c_2 = 0$ , (39) implies:

$$s_i = \frac{a - c_1}{3}, i = 1, 2 \tag{43}$$

During crowdfunding Firm 1 maximizes  $\pi((\frac{a+2c_1}{3})(\frac{a-c_1}{3}) - \beta c_1)$ . This gives us:

$$c_1 = \frac{a - 9\beta}{4} \tag{44}$$

Accordingly we have

$$\Pi_1 = \frac{\pi (a^2 - 2a\beta + 9\beta^2)}{8} \tag{45}$$

$$\Pi_{2} = \pi \left(\frac{a-c_{1}}{3}\right) \left(\frac{a-c_{1}}{3}\right) + (1-\pi)\left(\frac{a-c_{1}}{3}\right) \left(\frac{a+2c_{1}}{3}\right) =$$

$$= \frac{\pi (a+3\beta)^{2} + (1-\pi)(a+3\beta)(2a-3\beta)}{16}$$
(46)

**Proposition 6.** (1) If  $\beta > \frac{a}{9}$  or  $\pi < \frac{8a^2}{9(a^2-2a\beta+9\beta^2)}$ , the equilibrium is (S,S); (2) Otherwise the equilibrium is (CF,CF).

**Proof.** Consider the matrix of payoffs (Figure 4).

or

| Firm 1/Firm 2 | S   | CF  |
|---------------|---|---|
| S             | $\frac{a^2}{9}$ ; $\frac{a^2}{9}$                           | $\frac{\pi(a+3\beta)^2+(1-\pi)(a+3\beta)(2a-3\beta)}{16};$  |
|               |   | $\frac{\pi(a^2-2a\beta+9\beta^2)}{8}$   |
| CF            | $\frac{\pi(a^2-2a\beta+9\beta^2)}{8};$                      | $\frac{\pi(3a+a\pi-6\beta)(27a-18a\pi-a\pi^{2}+54\beta-6\beta\pi)}{4(9+\pi)^{2}}-\frac{\beta\pi(2a\pi-18\beta)}{9+\pi};$  |
|               | $\frac{\pi(a+3\beta)^2 + (1-\pi)(a+3\beta)(2a-3\beta)}{16}$ | $\left  \begin{array}{c} \frac{\pi (3a + a\pi - 6\beta)(27a - 18a\pi - a\pi^2 + 54\beta - 6\beta\pi)}{4(9 + \pi)^2} - \frac{\beta\pi (2a\pi - 18\beta)}{9 + \pi} \right $ |

Figure 4. Matrix of payoffs under risk of campaign failure.

First note that if  $\beta > \frac{a}{9}$  then according to (44) we have a corner solution with c = 0. Otherwise, the profit of Firm 1 in case (S,CF) is greater than it is in the case (S,S) if:

$$\frac{\pi(a^2 - 2a\beta + 9\beta^2)}{8} > \frac{a^2}{9}$$

$$\pi > \frac{8a^2}{9(a^2 - 2a\beta + 9\beta^2)}$$
(47)

This proves Part 1. As follows from the matrix of payoffs, whether (CF,CF) is an equilibrium, depends on the following:

$$\frac{\pi(3a+a\pi-6\beta)(27a-18a\pi-a\pi^2+54\beta-6\beta\pi)}{4(9+\pi)^2} - \frac{\beta\pi(2a\pi-18\beta)}{9+\pi} >$$
(48)  
$$> \frac{\pi(a+3\beta)^2 + (1-\pi)(a+3\beta)(2a-3\beta)}{16}$$

If this holds, the equilibrium is (CF,CF) and vice versa. In Section 4 we considered a similar case without risk of campaign failure, i.e., the case  $\pi = 1$ . In this case (48) becomes (12) which holds as we argued. In the general case (i.e., when  $0 < \pi < 1$ ), the numerical analysis of (48) for different values of  $\beta$  shows that when (47) holds, (CF,CF) is still an equilibrium. An example is illustrated in Figure 5. Since the payoff of Firm 1 from deviating from (CF,CF) is less then its equilibrium payoff, it will not deviate. The same argument applies for Firm 2.

This section adds to the analysis of the role of risk for crowdfunding. It crucially depends on the nature of the risk that the entrepreneur faces. If the risk is primarily related to demand uncertainty then as was shown in Proposition 4 the usage of crowdfunding is more beneficial because it helps the entrepreneur learn information about the demand. If however the risk is more related to the campaign problems then this risk reduces the likelihood of crowdfunding usage. Also note that some degree of risk might be beneficial. If the equilibrium is (CF,CF) then some degree of risk is beneficial i.e., further increase in the probability of success reduces firm profit in equilibrium as follows from Figure 5. Intuitions are similar to Section 7.2, i.e., some degree of risk of campaign failures may prevent firms from "stealing" inefficiently from their competitors by "playing" crowdfunding too much etc.

Previously we considered the case when the firm profit is zero in case when the campaign fails. Now suppose that campaign failure is costly for the entrepreneur. Although firm failure can provide a learning experience, there is always a risk of long-term reputation damage for entrepreneurs. In addition, it may lead to additional fees and costs if funders complain or sue the firm etc. Finally note that for example Indiegogo charges a higher fee for firms in the case when the campaign fails (for a discussion of consequences of failures see, for example, Greenberg and Gerber (2014)). If the failure is costy but information is symmetric between rivals, the previous analysis will not change. However, the cost of failure becomes important under asymmetric information because the thresholds (in case firms use AON) are different for different types of Firm 1 (e.g., high and low demand). More specifically suppose that if a firm fails during a crowdfunding campaign its net profit equals  $\Pi = pq - \beta c - B(c)$ ,

where *B* is the cost of failure, i.e., B(c) = b, b > 0 if c > 0 and the threshold is not reached and 0 otherwise.<sup>21</sup>



**Figure 5.** a = 1,  $\beta = 0.1$ . x-axis-probability of campaign success  $\pi$ . y-axis: solid line- firm payoff in case (CF,CF); dash line- Firm 1 payoff in case (S,CF).

Each type of firm can correctly anticipate its own demand and establish the threshold accordingly, however, when one firm type mimics the other type then the fact that the threshold values are different for each type may affect the firm's payoff if the cost of failure is sufficiently high and most importantly if the thresholds are publicly observable. In fact, the latter is usually the case.<sup>22</sup> For the most part, however, our results (Proposition 2) stand. Recall that in our equilibrium type *h* selects CF and type *l* selects S. The difference in thresholds will not help type *l* because *h* has a higher threshold. The opposite is true for the separating equilibrium where type *l* selects CF and type *h* selects S. This equilibrium does not exist because *h* will still mimic type *l*. This argument remains because the threshold for *l* is lower so when *h* mimics *l*, the threshold is still reached.

#### 7.5. Number of Backers

Here we assume that the number of units that can be sold during the pre-sale campaign (crowdfunding) is limited. For example, some people may not be aware of crowdfunding opportunities, some people may not like purchasing via the internet and/or do not trust online payment systems, some people may not have access to the internet etc. So let us assume that the demand during crowdfunding is limited:

$$c \le C, C > 0 \tag{49}$$

We argue that qualitatively, the predictions of our analysis do not change. For example, under symmetric information, (S,S) is not an equilibrium.

The matrix of payoffs for the different cases is shown in Figure 6 as we discuss below.

<sup>&</sup>lt;sup>21</sup> The calculations are very similar to Section 5 so they are omitted for brevity.

<sup>&</sup>lt;sup>22</sup> See, for example, https://www.kickstarter.com/projects/printdreams/printbrushtm-xdr-colors-everywhere?ref=sectionhomepage-featured-project-list

| Firm 1/Fir    | rm 2    S                                       |   | CF   |
|---------------|---|---|--|
| S             | $\frac{a^2}{9}; \frac{a^2}{9}$                  |   | $\frac{(a+3\beta)^2}{16}$ ; $\frac{a^2-2a\beta+9\beta^2}{8}$                   |
| CF            | $\left\  \frac{a^2-2a\beta+9\beta}{8} \right\ $ | $\frac{a^2}{16}; \frac{(a+3\beta)^2}{16}$ | $\frac{2a^2+4a\beta+27\beta^2}{25}$ ; $\frac{2a^2+4a\beta+27\beta^2}{25}$      |
|               |   | (a)                                       |  |
| Firm 1/Firm   | 2    S  |   | CF   |
| S             | $\frac{a^2}{9}; \frac{a^2}{9}$                  |   | $\frac{(a-C)^2}{9}; \frac{(a+2C)(a-C)}{9} - \beta C$                           |
| CF            | $\frac{(a+2C)(a-C)}{9}$                         | $-\beta C; \frac{(a-C)^2}{9}$             | $\frac{2a^2 + 4a\beta + 27\beta^2}{25}; \frac{2a^2 + 4a\beta + 27\beta^2}{25}$ |
|               |   | ( <b>b</b> )                              |  |
| Firm 1/Firm 2 | S   | C   | F  |
| S             | $\frac{a^2}{9}; \frac{a^2}{9}$                  | <u>(a</u>                                 | $(a-C)^2 = (a+2C)(a-C) = -\beta C$   |
| CF            | $\frac{(a+2C)(a-C)}{\alpha} - \beta C$          | $(a-C)^2$                                 | $\frac{(a+C)(a-2C)}{2} - \beta C$ ; $\frac{(a+C)(a-2C)}{2} - \beta C$          |

**Figure 6.** (a) Matrix of payoffs if  $C \ge \frac{a-9\beta}{4}$ ; (b) Matrix of payoffs if  $\frac{a-9\beta}{4} > C \ge \frac{a-9\beta}{5}$ ; (c) Matrix of payoffs if  $C < \frac{a-9\beta}{5}$ .

(c)

**Proposition 7.** If the number of potential backers is limited, (S,S) is still not an equilibrium.

Proof. If both firms select S, the equilibrium outcome is the same as it is in the unconstrained case.

Now consider a situation where both firms select CF. On the spot market, Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - s_2 - c_1 - c_2)$ , which makes:

$$s_1 = \frac{a - s_2 - c_1 - c_2}{2} \tag{50}$$

Similarly for Firm 2 we get

$$s_2 = \frac{a - s_1 - c_1 - c_2}{2} \tag{51}$$

Solving (50) and (51) produces

$$s_i = \frac{a - c_1 - c_2}{3} \tag{52}$$

During crowdfunding Firm 1 maximizes  $(s_1 + c_1)(a - s_1 - s_2 - c_1 - c_2) - \beta c_1$  subject to (52). This gives us:

$$c_1 = \min\{C, \frac{a-c_2-9\beta}{4}\}$$

Similarly for Firm 2:

$$c_2 = \min\{C, \frac{a-c_1-9\beta}{4}\}$$

It implies:

$$c_i = \frac{a - 9\beta}{5}$$
 if  $C \ge \frac{a - 9\beta}{5}$  and Cotherwise (53)

If  $C \geq \frac{a-9\beta}{5}$ , (52) and (53) imply that the price equals

$$p = \frac{a + 6\beta}{5}$$

The firm's profit then equals

$$\Pi_i = \frac{2a^2 + 4a\beta + 27\beta^2}{25}$$

If  $C < \frac{a-9\beta}{5}$ , (52) and (53) imply that

$$s_i = p = \frac{a - 2C}{3}$$

The firm's profit then equals

$$\Pi_i = \frac{(a+C)(a-2C)}{9} - \beta C$$

Now consider a situation where one firm (for instance, Firm 1) selects CF and the other selects S. In this case  $c_2 = 0$ .

Since  $c_2 = 0$ , (52) implies:

$$s_1 = s_2 = p = \frac{a - c_1}{3}$$
$$\Pi_1 = (c_1 + \frac{a - c_1}{3})\frac{a - c_1}{3} - \beta c_1 = \frac{(a + 2c_1)(a - c_1)}{9} - \beta c_1$$
(54)

Firm 2's profit equals

$$\Pi_2 = s_2 p = \frac{(a - c_1)^2}{9}$$

The optimal  $c_1$  that maximizes Firm 1's expected profit (given by (54)) equals

$$c_1 = \min\{\frac{a-9\beta}{4}, C\}\tag{55}$$

If  $C \ge \frac{a-9\beta}{4}$ ,  $c_1 = \frac{a-9\beta}{4}$  and accordingly  $\Pi_2 = \frac{(a+3\beta)^2}{16}$  and  $\Pi_1 = \frac{a^2-2a\beta+9\beta^2}{8}$ . If  $C < \frac{a-9\beta}{4}$ ,  $c_1 = C$ ,  $\Pi_2 = \frac{(a-C)^2}{9}$  and  $\Pi_1 = \frac{(a+2C)(a-C)}{9} - \beta C$ . First consider  $C \ge \frac{a-9\beta}{4}$ . The matrix of payoffs is the same as in Figure 1 so the proof follows from Proposition 1. Now let  $C < \frac{a-9\beta}{4}$ . Consider the situation where both firms select S. Suppose that

Firm 1 decides to use crowdfunding. As follows from Figure 6b,c Firm's 1 profit is greater than its profit under (S,S) if:

$$\frac{(a+2C)(a-C)}{9} - \beta C > \frac{a^2}{9}$$

which can be rewritten as

$$\frac{C(a-2C-9\beta)}{9} > 0$$

The latter holds if  $C < \frac{a-9\beta}{2}$ , which is the case.  $\Box$ 

Under asymmetric information the results remain very similar to the spirit of our basic model results. If C is very low, crowdfunding is not feasible and so no separating equilibrium exists. If C is sufficiently large then the results are the same as in Section 5. The reason is that if a firm can use crowdfunding in a separating equilibrium then all calculations (including the mimicking conditions by another firm) remain the same because when mimicking, the other firm should sell the same quantity during crowdfunding. So if this quantity is available to one firm it will also be available to another. The model can be further extended by assuming that a high-demand firm has a higher limit for crowdfunding than a low-demand firm. However, it does not change the signalling equilibrium analysis. Recall that our equilibrium is one where type *h* selects CF and type *l* selects S. The difference in limits on crowdfunding will not help type *l* because *h* has a higher limit. The opposite is true for the separating equilibrium where type *l* selects CF and type *h* selects S. This equilibrium does not exist because *h* will still mimic *l*. The argument remains because the limit for *l* is lower so when *h* mimics *l*, it can sell the same quantity since its limit is higher.

#### 7.6. The Distribution of Types

In the sections that deal with asymmetric information we use two types of firms to illustrate the main ideas. This is very typical in literature. A natural question though is whether the results stand if one considers a case with multiple types. Our analysis shows<sup>23</sup> that most conclusions remain the same: under asymmetric information, crowdfunding can often be used as a signal of high demand. In the case of multiple types, however, an equilibrium may exist where only the type with the lowest demand (speaking about Proposition 2 when  $\beta = 0$ ) is indifferent between crowdfunding and spot sales and every other type selects crowdfunding. When  $\beta > 0$ , our analysis shows that the results may hold even in a multiple types environment though more research is required. The main implication of our analysis holds. In particular, our results show that there is no semi-separating equilibrium where the average quality of types that choose crowdfunding is lower than those that choose spot sales, which is consistent with our basic model.

#### 7.7. Changing the Value of the Reward

One of the assumptions of our analysis is that the reward is just high enough to compensate buyers for waiting. Different concepts can instead be used. One should note, however, that without any additional assumptions ours is the simplest approach such that the non-arbitrage condition holds, which is one of the crucial assumptions when modelling consumer behavior and demand. Different papers make different assumptions in order to model rewards, for example in Belleflamme et al. (2014) there are ad-hoc community benefits related to crowdfunding which creates a difference between the pre-sale price and the spot price; in Chemla and Tinn (2019) a risk exists that the firm will not make any intermediate investment required to produce the product between the crowdfunding stage and the actual sales so the crowdfunding price may be lower than the spot price because of moral hazard and risk issues for funders etc.

#### 7.8. Different Types of Crowdfunding

Unlike capital structure literature, where a debt/equity mix is a very common strategy (as opposed to pure equity or pure debt financing), in our paper firms decide whether or not to use crowdfunding. So mixing these two decisions makes no sense. Further extensions, however, are possible where firms consider different types of crowdfunding.<sup>24</sup> We mostly used the KIA (mentioned previously) type of reward-based crowdfunding used by Indiegogo among others. However, the results are very close if instead one were to consider AON where firms have thresholds when conducting a crowdfunding campaign and when bankruptcy cost/cost of campaign failure are taken into consideration (without these costs no difference exists in principle since each company can rationally calculate the expected level of sales and/or crowdfunding profitability). The cost of failure becomes important under asymmetric information because the thresholds are different for the different types of Firm 1 (high and low demand). Each type of firm can correctly anticipate its own demand and establish the threshold accordingly, however, when one firm type mimics the other type then the fact that the threshold values are different for each type may affect the firm's payoff if the cost of failure is sufficiently high and most importantly if the thresholds are publicly observable. In fact, the latter is usually the case.<sup>25</sup> For the most part, however, our results (Proposition 2) stand. Recall that in our equilibrium type *h* selects CF and type *l* selects S. The difference in thresholds will not help type *l* 

<sup>&</sup>lt;sup>23</sup> Proofs are omitted for brevity. Note that the calculations become much longer and technically more complicated, which is very typical for multiple type games with asymmetric information.

<sup>&</sup>lt;sup>24</sup> For a literature on equity-based crowdfunding see, for example, Vulkan et al. (2016); Vismara (2016); Estrin et al. (2018); Mochkabadi and Volkmann (2018); Kleinert et al. (2020) or Miglo (2020). For debt-based crowdfunding see, for example, Kuo and Gerber (2012), Lenz (2016) and Kgoroeadira et al. (2019).

<sup>&</sup>lt;sup>25</sup> See, for example, https://www.kickstarter.com/projects/printdreams/printbrushtm-xdr-colors-everywhere?ref=sectionhomepage-featured-project-list

because *h* has a higher threshold, so it will become impossible for *l* to reach the required threshold and mimic *h*. As a result, an equilibrium where type *h* selects CF will still exist. The opposite is true for the separating equilibrium where type *l* selects CF and type *h* selects S. This equilibrium does not exist because *h* will still mimic type *l*. This argument remains because the threshold for *l* is lower so when *h* mimics *l*, the threshold will still be reached.

Note that existing theoretical literature on crowdfunding often considers reward-based crowdfunding separately from equity-based crowdfunding and debt-based crowdfunding. One of the reasons for this seems to be that the founders' objectives are quite different in these scenarios (see, for example, Hildebrand et al. (2016)). This is definitely an interesting direction for future research.

#### 7.9. Risk-Averse Entrepreneurs

In our model all decision-makers are risk-neutral. This is in keeping with traditional capital structure literature (beginning with Modigliani and Miller (1958)) where, with a few exceptions (e.g., Leland and Pyle (1977)) risk aversion does not play a major role. This is in part because firms can manage their risk by adjusting their capital structures, investors can manage their risk by adjusting their home-leverage, and shareholders can minimize their risk by sharing risk with other shareholders etc. In entrepreneurial finance literature entrepreneurial risk-aversion is a peculiar idea since entrepreneurs by definition are supposed to be risk-takers (see, for example, Villot (2011)). Our preliminary analysis shows that our basic results (e.g., Proposition 1) will not change if entrepreneurs are risk-averse but calculations will become much longer. This can be a direction for future reserach.

The results of our analysis and teastable empirical predictions are summarized in Table 2.

| Table 2. The model's results and testable empirical predictions |
|---|
|---|

| The Model's Predictions   |
|---|
| Competition increases the chances of using crowdfunding compared to the monopoly case                 |
| Firms can use crowdfunding strategically to signal a high level of demand for their products          |
| The likelihood of crowdfunding increases with demand uncertainty                                      |
| Higher cost of waiting/rate of discounting decreases the likelihood of crowdfunding                   |
| Higher platform fee (per value) may lead to higher firm profits in equilibrium                        |
| The amount of (reward-based) crowdfunding is pro-cyclical   |
| A non-monotonic relationship exists between the risk of crowdfunding campaign failure and firm profit |
|   |

## 8. Conclusions

Crowdfunding is a growing area of interest among practitioners and theorists (see, e.g., Ahlstrom et al. (2018)). It is especially important for small businesses, innovative businesses and start-up businesses because for them financing is a crucial factor of success. Unlike large or established businesses, they do not typically have a long and successful credit history and do not own a large amount of tangible assets in order to successfully negotiate traditional forms of financing such as bank loans. It is not surprising then that these firms are often considered pioneers in finding new ways of funding their projects including private equity in the 1940s, business incubators in the 1950s, venture capital finance in the 1960s, angel finance in the 1970s and seed accelerators in the 2000s. Crowdfunding in its modern form (performed online) is one of the newest ideas in this area and is sometimes credited as a top 10 invention of the 21st century.<sup>26</sup> In the last 10-15 years, crowdfunding has been playing a significant role in enhancing entrepreneurial potential for innovation and economic growth (Belleflamme et al. (2015); Estrin et al. (2018)).

<sup>&</sup>lt;sup>26</sup> http://www2.technologyreview.com/tr10/?year=2012

Existing literature has discovered many reasons for why raising funds is difficult for entrepreneurial companies including asymmetries of information, high bankruptcy costs, transactions costs, moral hazard issues and many others. Our paper contributes to the growing literature on crowdfunding in particular to the branch that focuses on the role of crowdfunding in mitigating asymmetric information problems. Theoretical literature on crowdfunding usually considers a monopoly setting. It is not surprising given that crowdfunding was originally used to fund highly innovative projects where firms retain monopoly power during the development and sale of the given products. Recently, however, companies have begun to use crowdfunding to finance more traditional products and services where they compete against other sellers of similar products. Our main contribution is that we consider the role of crowdfunding in a competitive setting.<sup>27</sup> This paper offers a model of a duopoly where firms can use crowdfunding prior to direct sales. The model is based on asymmetric information between competitors regarding the demand for the product.

We first demonstrate that in a monopoly setting, without making any additional assumptions such as community benefits to funders, crowdfunding is never used and firms prefer to just use the spot market since crowdfunding involves a cost in the form of rewards. Then we consider a duopoly.

Our paper also contributes to the literature that analyzes different marketing tools for a firm operating in a competitive environment. This encompasses forward trading, advanced sales, social coupons, price discrimination etc. It is known that competition weakens or eliminates the effectiveness of many marketing strategies (e.g., bundling, price discrimination). In our paper, competition does not diminish the advantage of crowdfunding (comparing duopoly and monopoly cases for example) because, unlike price discrimination, consumer reward applies to all consumers in the pre-sale period so one seller is unable to focus his/her attention on only one group of consumers (i.e., those being discriminated against).

In Sections 4 and 5 both competition and imperfect information including information asymmetry are present and their interaction and implications for crowdfunding decisions are analyzed. As was mentioned previously, existing theoretical literature focuses on crowdfunding in a monopoly setting. The main deficiency of this line of research and its value for entrepreneurial firms is that it undermines the importance of competition. An increasingly important aspect of competition is information management, which includes, on one hand, acquiring information about competitors and on the other hand managing a firm's own information and developing a strategy of sharing its own information with the outside world etc. (see, among others, Dixon (1992); Lubit (2001); Wang and Ellinger (2011); Jeitschko et al. (2016) and Chen et al. (2017)). Furthermore, at the highest levels of competition, firms may have an incentive to deliberately distort their actions to undermine the ability of their rivals to make inferences about their underlying private information (Jeitschko et al. (2016)). We argue that crowdfunding can play a strategic role in competitive markets. For example we find that high-demand firms can use crowdfunding to signal their quality. Hence our paper also contributes to the literature on oligopolistic competition under asymmetric information. As was previously mentioned, this literature has analyzed many issues including, for example, firm strategy regarding information acquisition, consumer satisfaction, confidentiality and the social value of information but not in regards to crowdfunding. Crowdfunding literature dicusses some asymmetric information problems in a monopoly setting but it has not analyzed the role of asymmetric information between rivals.

Our paper also contributes to the capital structure literature that connects firm financing decisions and production decisions in a competitive setting (for a recent review see, for example, Li and Wang (2019)). Our paper is the first to introduce reward-based crowdfunding as a financing choice for a firm. We also analyze the role of asymmetric information. Both of these aspects (crowdfunding and asymmetric infromation) have not yet been significantly considered in papers in this field of capital

<sup>&</sup>lt;sup>27</sup> In a similar spirit, to some extent, Dushnitsky and Fitza (2018) suggest that crowdfunding research should focus more on comparing different crowdfunding platforms and respectively on analyzing competition between platforms.

structure analysis (see, for example, Brander and Lewis (1986); Bolton and Scharfstein (1990) and Williams (1995)). As discussed in Section 6, we leave many interesting topics for future research, such as: considering heterogeneity of consumers' preferences and other modifications of the demand function, introducing different types of crowdfunding and other types of financing into the model, and considering dynamic extensions of the model.

Funding: This research received no external funding.

Conflicts of Interest: The author declares no conflicts of interest.

#### Appendix A

# Appendix A.1. Proof of Proposition 2

Consider a situation where Firm 2 selects CF and Firm 1 selects CF only if it is type l and selects S otherwise. Firm 2 believes that Firm 1's type is l when observing CF and h when observing S. From Figure 2 the payoffs of each type are

$$\Pi_{1l} = \frac{2a_l^2}{25} + \frac{4a_l\beta}{25} + \frac{27\beta^2}{25}$$
(A1)

$$\Pi_{1h} = \frac{a_h^2}{16} + \frac{6a_h\beta}{16} + \frac{9\beta^2}{16} \tag{A2}$$

where  $\Pi_{1j}$  is the equilibrium profit of type *j* (all calculations are based on the symmetric information case for each type described in Section 4. Suppose that *h* mimics *l* and uses crowdfunding. (41) implies

$$c_2 = \frac{a_l}{5} - \frac{9\beta}{5}$$

After learning information about demand, Firm 2 chooses  $s_2$  to maximize  $s_2(a_h - s_1 - s_2 - c_1 - c_2)$ , which makes:

$$s_2 = \frac{5a_h - 5s_1 - 5c_1 - a_l + 9\beta}{10} \tag{A3}$$

Similarly for Firm 1 we get

$$s_1 = \frac{5a_h - 5s_2 - 5c_1 - a_l + 9\beta}{10} \tag{A4}$$

Solving (A3) and (A4) produces

$$s_i = \frac{a_h - c_1 - \frac{a_l - 9\beta}{5}}{3} = \frac{5a_h - 5c_1 - a_l + 9\beta}{15}$$

Also

$$p = a_h - 2\frac{5a_h - 5c_1 - a_l + 9\beta}{15} - c_1 - \frac{a_l - 9\beta}{5} = \frac{5a_h - 5c_1 - a_l + 9\beta}{15}$$

During crowdfunding Firm 1 maximizes  $\left(\frac{5a_h-5c_1-a_l+9\beta}{15}+c_1\right)\frac{5a_h-5c_1-a_l+9\beta}{15}-\beta c_1$ . This gives us:

$$c_1 = \frac{5a_h - a_l + 9\beta}{20}$$
$$p = \frac{5a_h - a_l + 9\beta}{20}$$

Firm 1's profit then equals

$$\Pi_{1h} = \left(\frac{5a_h - a_l + 9\beta}{20} + \frac{5a_h - a_l + 9\beta}{20}\right)\frac{5a_h - a_l + 9\beta}{20} = \frac{(5a_h - a_l + 9\beta)^2}{200}$$

Comparing this with (A2) we find that it is smaller if

$$4a_l + 6(5\sqrt{2} - 6)\beta > 10(2 - \sqrt{2})a_h \tag{A5}$$

This never holds. Indeed, the maximum value of left side is reached when  $\beta = a_l/9$ . In this case (A5) becomes:  $\frac{5\sqrt{2}}{3}a_l > 5(2-\sqrt{2})a_h$  or  $\frac{\sqrt{2}}{3(2-\sqrt{2})}a_l > a_h$ . This does not hold because  $\frac{\sqrt{2}}{3(2-\sqrt{2})} < 1$  and  $a_l < a_h$ . So this equilibrium does not exist. If Firm 2 perceives a firm that uses a crowdfunding campaign to be a low demand firm, it will be pessimistic about the price and will concede the market. Type *h* can benefit from this situation by mimicking the strategy of type *l* during the crowdfunding stage.

The second part of the proposition follows from the case  $\beta = 0$  discussed in the main text. Since an equilibrium exists when  $\beta = 0$ , it also exists when  $\beta$  is sufficiently small because the non-mimicking conditions hold as strict inequalities and profit functions are continuous in  $\beta$  (see also Miglo (2018)).

#### Appendix A.2. Pooling Equilibrium Analysis

Our analysis shows that the only candidate for a pooling equilibrium is one where both types of Firm 1 select CF and Firm 2 selects CF as well.<sup>28</sup> To illustrate the existence of this equilibrium consider the case  $\beta = 0$ . Off-equilibrium beliefs of Firm 2 when observing S is that it is type h.<sup>29</sup> Let  $c_{1j}$  and  $s_{1j}$  denote the volume of crowdfunding and spot sales by the type j of Firm 1, j = l, h. In contrast to a signalling equilibrium in the case of a pooling equilibrium, Firm 2 will update its beliefs about Firm 1's type at the end of the pre-sale period after observing the volume of crowdfunding. If Firm 2 discovers that it has a high-demand product, then on the spot market, Firm 2 chooses  $s_2$  to maximize  $s_2(a_h - s_{1h} - s_2 - c_{1h} - c_2)$ , which makes:

$$s_2 = \frac{a_h - s_{1h} - c_{1h} - c_2}{2} \tag{A6}$$

Similarly for Firm 1 we get

$$s_{1h} = \frac{a_h - s_2 - c_{1h} - c_2}{2} \tag{A7}$$

Solving (A6) and (A7) produces

$$s_2 = s_{1h} = \frac{a_h - c_{1h} - c_2}{3} \tag{A8}$$

Similarly if Firm 2 discovers it has a low-demand product we have

$$s_2 = s_{1l} = \frac{a_l - c_{1l} - c_2}{3} \tag{A9}$$

During crowdfunding Firm 2 chooses  $c_2$  to maximize  $\mu \Pi_h(c_2) + (1 - \mu)\Pi_l(c_2)$ , where  $\Pi_h(c_2) = (s_2 + c_2)(a_h - s_{1h} - s_2 - c_{1h} - c_2)$  subject (A8) and  $\Pi_l(c_2) = (s_2 + c_2)(a_l - s_{1l} - s_2 - c_{1l} - c_2)$  subject to (A9). This means that Firm 2 maximizes its expected profit given that the probability that Firm 1 is type *h* is  $\mu$ . This gives us:

$$c_2 = \frac{\mu a_h + (1-\mu)a_l - \mu c_{1h} - (1-\mu)c_{1l}}{4}$$

Similarly for Firm 1 (if type *h*):

$$c_{1h} = \frac{a_h - c_2}{4}$$

<sup>&</sup>lt;sup>28</sup> The formal proof is omitted for brevity.

<sup>&</sup>lt;sup>29</sup> Intuitively, it is because Firm 2 is more aggressive when it believes that the type is h (high demand). In this case it increases sales which ultimately reduces its competitor's profit. It makes any deviations by Firm 1 less attractive.

J. Risk Financial Manag. 2020, 13, 39

and

$$c_{1l} = \frac{a_l - c_2}{4}$$

if type *l*. It implies:

$$c_2 = \frac{\mu a_h + (1 - \mu)a_l}{5} \tag{A10}$$

$$c_{1h} = \frac{(5-\mu)a_h - (1-\mu)a_l}{20} \tag{A11}$$

$$c_{1l} = \frac{(4+\mu)a_l - \mu a_h}{20}$$

(A8), (A10) and (A11) imply (for type *h*)

$$s_2 = s_{1h} = p_h = \frac{(5-\mu)a_h - (1-\mu)a_h}{20}$$

and for type *l* we have:

$$s_2 = s_{1l} = p_l = \frac{(4+\mu)a_l - \mu a_h}{20}$$

Firm 1's (type *h*) profit equals then

$$\Pi_{1h} = \frac{((5-\mu)a_h - (1-\mu)a_l)^2}{200}$$
(A12)

$$\Pi_{1l} = \frac{((4+\mu)a_l - \mu a_h)^2}{200}$$
(A13)

$$\Pi_{2} = \mu \Pi_{h}(c_{2}) + (1-\mu)\Pi_{l}(c_{2}) = \mu \left(\frac{(5-\mu)a_{h} - (1-\mu)a_{l}}{20} \left(\frac{\mu a_{h} + (1-\mu)a_{l}}{5} + \frac{(5-\mu)a_{h} - (1-\mu)a_{l}}{20}\right) + (1-\mu)\left(\frac{(4+\mu)a_{l} - \mu a_{h}}{20} \left(\frac{\mu a_{h} + (1-\mu)a_{l}}{5} + \frac{(4+\mu)a_{l} - \mu a_{h}}{20}\right)\right)$$

Suppose that *h* deviates and does not use crowdfunding. The optimal strategy for Firm 2 is to use CF. So Firm 1's profit is  $\frac{a_h^2}{16}$ . Comparing this with (A12) we find that *h* will deviate if

$$\frac{10a_h - 5\sqrt{2}a_h - 2a_l}{2(a_h - a_l)} < \mu < \frac{10a_h + 5\sqrt{2}a_h - 2a_l}{2(a_h - a_l)}$$

Since  $\frac{10a_h - 5\sqrt{2}a_h - 2a_l}{2(a_h - a_l)} > 1$ , it does not hold so *h* does not deviate. Now suppose that *l* deviates and does not use crowdfunding. The calculations are similar to the

Now suppose that l deviates and does not use crowdfunding. The calculations are similar to the case of l mimicking h in the case of the signalling equilbrium discussed previously. According to (23)

$$\Pi_{1l} = \frac{(11a_h - 8a_l)(4a_l - a_h)}{144}$$

Comparing this with (A13) we find that l will not deviate if

$$\mu < \frac{24a_l - 5\sqrt{2(11a_h - 8a_l)(4a_l - a_h)}}{6(a_h - a_l)} \tag{A14}$$

Now we check that off-equilibrium beliefs satisfy the Cho-Kreps intuitive criterion. We will demonstrate that for type h its payoff from deviation is greater than its equilibrium payoff when Firm 2 perceives it to be type l. In this case Firm 2 uses CF and the profit of type h is (similar to previous calculations):

$$\Pi_{1h} = \frac{(2a_h - a_l)^2}{16}$$

This is greater than its equilbrium payoff when

$$\mu > \frac{a_l(5\sqrt{2}-2) - 10a_h(\sqrt{2}-1)}{2(a_h - a_l)}$$
(A15)

In this case off-equilibrium beliefs survive the Cho-Kreps intuitive criterion.

Note that conditions (A14) and (A15) can hold simultaneously. Consider, for example, the case  $a_l = 1$  and  $a_h = 3.98$ .

Therefore a pooling equilibrium where both types of Firm 1 select crowdfunding may exist. The second part of the proposition was considered in the main text.

#### Appendix A.3. Proof of Proposition 4

For brevity, the case  $\mu = 0.5$  is considered and the outline of the proof is provided. Full calculations are similar to Proposition 1 and are available upon request. In stage 1 the following situations can occur: both firms select S; both firms select CF; Firm 1 selects CF and Firm 2 selects S; Firm 2 selects CF and Firm 1 selects S. If both firms select S, the equilibrium outcome is  $s_1 = s_2 = \frac{a_h + a_l}{6}$  and  $\Pi_1 = \Pi_2 = \frac{(a_h + a_l)^2}{36}$ . Indeed Firm 1 chooses  $s_1$  to maximize  $0.5(a_h - s_1 - s_2)s_1 + 0.5(a_l - s_1 - s_2)s_1$ , which makes:

$$s_1 = \frac{0.5a_h + 0.5a_l - s_2}{2} \tag{A16}$$

Similarly for Firm 2 we get

$$s_2 = \frac{0.5a_h + 0.5a_l - s_1}{2} \tag{A17}$$

Solving (A16) and (A17) produces

$$s_i = \frac{a_h + a_l}{6}, i = 1, 2$$
  
 $\Pi_i = \frac{(a_h + a_l)^2}{36}, i = 1, 2$ 
(A18)

Also

Now consider a situation where both firms select CF. We begin the solution by working backwards. On the spot market, Firm 1 chooses  $s_1$  to maximize  $s_1(a - s_1 - c_1 - (s_2 + c_2))$ , which makes:

$$s_1 = \frac{a - c_1 - (s_2 + c_2)}{2} \tag{A19}$$

Similarly for Firm 2 we get

$$s_2 = \frac{a - c_2 - (s_1 + c_1)}{2} \tag{A20}$$

Solving (A19) and (A20) produces

$$s_1 = \frac{a - 2c_1 + c_1 - c_2}{3} \tag{A21}$$

$$s_2 = \frac{a - 2c_2 + c_2 - c_1}{3} \tag{A22}$$

Also

$$p = a - c_1 - c_2 - s_1 - s_2 = \frac{a - c_1 - c_2}{3}$$

So if after the crowdfunding stage firms discover that  $a = a_h$ , we have  $s_{1h} = s_{2h} = p_h = \frac{a_h - c_1 - c_2}{3}$ . Similarly for the case  $a = a_l$ , we have  $s_{1l} = s_{2l} = p_l = \frac{a_l - c_1 - c_2}{3}$ .

During crowdfunding Firm 1 maximizes  $0.5(p_h(c_1 + s_{1h}) - \beta c_1) + 0.5(p_l(c_1 + s_{1l}) - \beta c_1) = 0.5(\frac{a_h - c_1 - c_2}{3} - \beta c_1) + 0.5(\frac{a_l - c_1 - c_2}{3} - \beta c_1) + 0.5(\frac{a_l - c_1 - c_2}{3} - \beta c_1)$ . This gives us:

J. Risk Financial Manag. 2020, 13, 39

$$c_1 = \frac{a_h + a_l - 2c_2 - 18\beta}{8}$$

Similarly for Firm 2:

$$c_2 = \frac{a_h + a_l - 2c_1 - 18\beta}{8}$$

This implies:

$$c_i = \frac{a_h + a_l - 18\beta}{10}, i = 1, 2 \tag{A23}$$

(A21)–(A23) imply that the prices equal

$$p_i = \frac{a+6\beta}{5}, i = 1, 2$$

The firms' expected profits then equal

$$\Pi_{i} = 0.5\left(\frac{a_{h} - c_{1} - c_{2}}{3}\frac{a_{h} + 2c_{1} - c_{2}}{3} - \beta c_{1}\right) + 0.5\left(\frac{a_{l} - c_{1} - c_{2}}{3}\frac{a_{l} + 2c_{1} - c_{2}}{3} - \beta c_{1}\right) = \frac{43a_{h}^{2} + 43a_{l}^{2} - 14a_{h}a_{l} + 72a_{l}\beta + 72a_{h}\beta + 972\beta^{2}}{900}, i = 1, 2$$

Now consider a situation where one firm (for instance, Firm 1) selects CF and the other one selects S. In this case  $c_2 = 0$ .

Since  $c_2 = 0$ , (A21) and (A22) imply:

$$s_1 = \frac{a - c_1}{3}$$
 (A24)

$$s_2 = \frac{a - c_1}{3} \tag{A25}$$

Also

$$p = a - c_1 - s_1 - s_2 = \frac{a - c_1}{3}$$

So if after the crowdfunding stage firms discover that  $a = a_h$ , we have  $s_{1h} = s_{2h} = p_h = \frac{a_h - c_1}{3}$ . Similarly for the case  $a = a_l$ , we have  $s_{1l} = s_{2l} = p_l = \frac{a_l - c_1}{3}$ . During crowdfunding Firm 1 maximizes  $0.5(p_h(c_1 + s_{1h}) - \beta c_1) + 0.5(p_l(c_1 + s_{1l}) - \beta c_1)$ . This gives us:

$$c_1 = \frac{a_h + a_l - 18\beta}{8}$$
(A26)

Accordingly we have

$$\Pi_{1} = \frac{17a_{h}^{2} + 17a_{l}^{2} + 2a_{h}a_{l} - 90a_{l}\beta - 90a_{h}\beta + 648\beta^{2}}{288}$$
$$\Pi_{2} = 0.5s_{2}p + 0.5s_{2}p = 0.5(\frac{7a_{h} - a_{l} + 18\beta}{24})^{2} + 0.5(\frac{7a_{l} - a_{h} + 18\beta}{24})^{2} = \frac{25a_{h}^{2} + 25a_{l}^{2} - 14a_{h}a_{l} + 108a_{l}\beta + 108a_{h}\beta + 162\beta^{2}}{576}$$

The above analysis leads to the following matrix of payoffs.

| Firm 1/Firm 2 | S  | CF   |
|---------------|--|--|
| S             | $\frac{(a_h+a_l)^2}{36}$ ; $\frac{(a_h+a_l)^2}{36}$  | $\frac{25a_h^2 + 25a_l^2 - 14a_ha_l + 108a_l\beta + 108a_h\beta + 324\beta^2}{576};$ |
|               |  | $\frac{17a_h^2 + 17a_l^2 + 2a_ha_l - 36a_l\beta - 36a_h\beta + 324\beta^2}{288}$     |
| CF            | $\left  \frac{17a_h^2 + 17a_l^2 + 2a_ha_l - 36a_l\beta - 36a_h\beta + 324\beta^2}{288}; \right $ | $\frac{43a_h^2+43a_l^2-14a_ha_l+72a_l\beta+72a_h\beta+972\beta^2}{900};$             |
|               | $\frac{25a_h^2 + 25a_l^2 - 14a_ha_l + 108a_l\beta + 108a_h\beta + 324\beta^2}{576}$              | $\frac{43a_h^2 + 43a_l^2 - 14a_ha_l + 72a_l\beta + 72a_h\beta + 972\beta^2}{900}$    |

Figure A1. Matrix of payoffs with uncertain demand.

To illustrate Proposition 4 we consider two cases ("small beta" and "large beta"). Consider first the case  $\beta = 0$ . The matrix of payoffs becomes:

| Firm 1/Firm 2 | S   | CF   |
|---------------|---|--|
| S             | $\frac{(a_h+a_l)^2}{36}; \frac{(a_h+a_l)^2}{36}$                                    | $\left \left \frac{25a_{h}^{2}+25a_{l}^{2}-14a_{h}a_{l}}{576};\frac{17a_{h}^{2}+17a_{l}^{2}+2a_{h}a_{l}}{288}\right \right $ |
| CF            | $\frac{17a_h^2 + 17a_l^2 + 2a_ha_l}{288}; \frac{25a_h^2 + 25a_l^2 - 14a_ha_l}{576}$ | $\frac{43a_h^2+43a_l^2-14a_ha_l}{900}; \frac{43a_h^2+43a_l^2-14a_ha_l}{900}$   |

**Figure A2.** Matrix of payoffs with uncertain demand and  $\beta = 0$ .

(CF,CF) is an equilibrium. Indeed it follows from Figure A2 that switching to crowdfunding from the case (S,S) improves the payoff of Firm 1. This is because  $\frac{17a_h^2 + 17a_l^2 + 2a_ha_l}{288} > \frac{(a_h + a_l)^2}{36}$ . This can be written as

$$\frac{9(a_h - a_l)^2 + 4a_h a_l}{288} > 0$$

which holds. So (S,S) is not an equilibrium. Whether the equilibrium is (S,CF) or (CF,CF) depends on the following:

$$\frac{43a_h^2 + 43a_l^2 - 14a_ha_l}{900} > \frac{25a_h^2 + 25a_l^2 - 14a_ha_l}{576}$$
(A27)

If this holds, the equilibrium is (CF,CF) and vice versa. (A27) can be written as

$$\frac{7(a_h - a_l)^2}{6400} > 0$$

which holds so (CF,CF) is an equilibrium. On the other hand consider the case  $\beta = 0.1$ . The matrix of payoffs is now:

| Firm 1/Firm 2    S   | CF  |
|--|---|
| S $\left  \frac{(a_h+a_l)^2}{36}; \frac{(a_h+a_l)^2}{36} \right $      | $\left  \frac{25a_h^2 + 25a_l^2 - 14a_ha_l + 10.8a_l + 10.8a_h + 3.24}{576}; \right $ |
|  | $\frac{17a_h^2 + 17a_l^2 + 2a_ha_l - 3.6a_l - 3.6a_h + 3.24}{288}$                    |
| CF $\frac{17a_h^2 + 17a_l^2 + 2a_ha_l - 3.6a_l - 3.6a_h + 3.24}{288};$ | $\frac{43a_h^2+43a_l^2-14a_ha_l+7.2a_l+7.2a_h+9.72}{900};$                            |
| $\begin{array}{                                    $                   | $\frac{43a_h^2 + 43a_l^2 - 14a_ha_l + 7.2a_l + 7.2a_h + 9.72}{900}$                   |

**Figure A3.** Matrix of payoffs with uncertain demand and  $\beta = 0.1$ .

Similar calculations show that: (1) If  $a_h = a_l = 0.9$  (S,S) is an equilibrium; (2) If  $a_h = 1.8$  and  $a_l = 0$ , equilibrium is (CF,CF). Note that the average demand is the same as previously but uncertainty is much higher now.

#### Appendix A.4. Cournot Duopoly with Cost

This subsection presents the results of the analysis of the model in Section 4 with cost. A firm's profit equals  $p(c_i + s_i) - (c_i + s_i)^2/2 - \beta c_i$ , i = 1, 2, where  $(c_i + s_i)^2/2$  is the production cost. The equilibrium concept and the sequence of events are exactly the same as they are in Section 4. The calculations lead to the following proposition.<sup>30</sup>

**Proposition A1.** Under symmetric information and known demand: (1) If  $\beta > \frac{a}{32}$ . the equilibrium is (S,S); (2) Otherwise the equilibrium is (CF,CF). This equilibrium is Pareto-inefficient. Under symmetric information and unknown demand: (1) If  $\beta$  is sufficiently small, the equilibrium is (CF,CF); (2) Otherwise the equilibrium is (S,S); (3) for a given value  $\beta$ , the equilibrium is (CF,CF) if the difference between  $a_h$  and  $a_1$  is large enough. Otherwise, the equilibrium is (S,S). Under asymmetric information: (1) A separating equilibrium where type l selects CF and type h selects S does not exist; (2) if  $\beta$  is sufficiently small, a separating equilibrium exists where type h selects CF and type l selects S.

To illustrate the proposition consider  $\beta < \frac{a}{32}$ . The matrix of payoffs is below.

| Firm 1/Firm 2 | S  | CF   |
|---------------|--|--|
| S             | $\frac{3a^2}{32}; \frac{3a^2}{32}$                                   | $\frac{(5a+8\beta)^2}{294};\frac{2a^2-2a\beta+32\beta^2}{21}$                          |
| CF            | $\frac{2a^2 - 2a\beta + 32\beta^2}{21}; \frac{(5a + 8\beta)^2}{294}$ | $\frac{21a^2 + 42a\beta + 384\beta^2}{242}; \frac{21a^2 + 42a\beta + 384\beta^2}{242}$ |

Figure A4. Matrix of payoffs in Cournot duopoly with cost.

First note that as follows from the matrix of payoffs, switching to crowdfunding from the case (S,S) improves the payoff of Firm 1. So (S,S) is not an equilibrium. Indeed the profit of Firm 1 in case (S,CF) is greater than it is in case (S,S) if:

$$\frac{2a^2 - 2a\beta + 32\beta^2}{21} > \frac{3a^2}{32}$$

This holds because it can be written as

$$\frac{(a-32\beta)^2}{672} > 0$$

As follows from the matrix of payoffs, whether the equilibrium is (S,CF) or (CF,CF) depends on the following:

$$\frac{21a^2 + 42a\beta + 384\beta^2}{242} > \frac{(5a + 8\beta)^2}{294}$$
(A28)

If this holds, the equilibrium is (CF,CF) and vice versa. (A28) can be written as

$$\frac{(a-32\beta)(31a-761\beta)}{17787} > 0$$

which holds because  $a > 32\beta$  so (CF,CF) is an equilibrium. Pareto-efficiency of (CF,CF) depends on the following:

$$\frac{21a^2 + 42a\beta + 384\beta^2}{242} > \frac{2a^2 - 2a\beta + 32\beta^2}{21}$$
(A29)

<sup>&</sup>lt;sup>30</sup> Full calculations are very similar to Section 4 so they are omitted for brevity.

If this holds, (CF,CF) is Pareto-efficient and vice versa. (A29) can be written as

$$\frac{(a-32\beta)(-10\beta-43a)}{200} > 0$$

The left side of this inequality is negative because  $a > 32\beta$ . So (CF,CF) is not Pareto-efficient.

#### References

- Agrawal, Ajay, Christian Catalini, and Avi Goldfarb. 2015. Crowdfunding: Geography, Social Networks, and the Timing of Investment Decisions (Summer 2015). *Journal of Economics & Management Strategy* 24: 253–74. [CrossRef]
- Ahlers, Gerrit K. C., Douglas J. Cumming, Christina Guenther, and Denis Schweizer. 2015. Signaling in Equity Crowdfunding. *Entrepreneurship Theory and Practice* 39: 955–80. [CrossRef]
- Alegre, Ines, and Melina Moleskis. Crowdfunding: A Review and Research Agenda (October 7, 2016). Working Paper. Available online: http://dx.doi.org/10.2139/ssrn.2900921 (accessed on 22 February 2020).
- Ahlstrom, David, Douglas Cumming, and Silvio Vismara. 2018. New methods of entrepreneurial firm financing: Fintech, crowdfunding and corporate governance implications. *Corporate Governance: An International Review* 26: 310–13. [CrossRef]
- Allaz, Blaise. 1992. Oligopoly, Uncertainty and Strategic Forward Transactions. *International Journal of Industrial* Organization 10: 297–308. [CrossRef]
- Askew, Peter. 2016. Digital Farmers Market Aims to Disrupt the Duopoly by Directly Supporting Aussie Farmers. Available online: https://www.linkedin.com/pulse/digital-farmers-market-aims-disrupt-duopoly-directlysupporting (accessed on 22 February 2020).
- Beaulieu, Tanya, Suprateek Sarker, and Saonee Sarker. 2015. A Conceptual Framework for Understanding Crowdfunding. *Communications of the Association for Information Systems* 37: 1–31. [CrossRef]
- Belleflamme, Paul, Thomas Lambert, and Armin Schwienbacher. 2014. Crowdfunding: Tapping the Right Crowd. Journal of Business Venturing: Entrepreneurship, Entrepreneurial Finance, Innovation and Regional Development 29: 585–609. [CrossRef]
- Belleflamme, Paul, Nessrine Omrani, and Martin Peitz. 2015. The Economics of Crowdfunding Platforms. *Information Economics and Policy* 33: 11–28. [CrossRef]
- Bolton, Patrick, and David Scharfstein. 1990. A Theory of Predation Based on Agency Problems in Financial Contracting. *American Economic Review* 80: 93–106.
- Brander, James, and Tracy Lewis. 1986. Oligopoly and Financial Structure: The Limited Liability Effect. *American Economic Review* 76: 956–70.
- Burtch, Gordon, Anindya Ghose, and Sunil Wattal. 2013. An empirical examination of the antecedents and consequences of contribution patterns in crowd-funded markets. *Information Systems Research* 24: 499–519. [CrossRef]
- Chakraborty, Soudipta, and Robert Swinney. 2019. Signalling to the Crowd: Private Quality Information and Rewards-Based Crowdfunding. *Manufacturing and Service Operations Management* Forthcoming. Available online: http://dx.doi.org/10.2139/ssrn.2885457 (accessed on 22 February 2020).
- Chaney, Damien. 2019. A principal–agent perspective on consumer co-production: Crowdfunding and the redefinition of consumer power. *Technological Forecasting and Social Change* 141: 74–84. [CrossRef]
- Chang, Jen-Wen. 2016. The Economics of Crowdfunding (August 30, 2016). Available online: https://ssrn.com/ abstract=2827354 (accessed on 22 February 2020).
- Chemla, Gilles, and Katrin Tinn. 2019. Learning Through Crowdfunding. *Management Science* Forthcoming. [CrossRef]
- Chen, Tianxu, Mark Tribbitt, Yi Yang, and Xiaomei Li. 2017. Does rivals' innovation matter? A competitive dynamics perspective on firms' product strategy. *Journal of Business Research* 76: 1–7. [CrossRef]
- Cho, In-Koo, and David Kreps. 1987. Signaling Games and Stable Equilibria. *The Quarterly Journal of Economics* 102: 179–221. [CrossRef]
- Cholakova, Magdalena, and Bart Clarysse. 2015. Does the Possibility to Make Equity Investments in Crowdfunding Projects Crowd Out Reward-Based Investments? *Entrepreneurship Theory and Practice* 39: 145–72. [CrossRef]

- Cordova, Alessandro, Johanna Dolci, and Gianfranco Gianfrate. 2015. The Determinants of Crowdfunding Success: Evidence from Technology Projects'. *Procedia Social and Behavioral Sciences* 181: 115–24. [CrossRef]
- Cumming, Douglas, and Lars Hornuf. 2018. *The Economics of Crowdfunding Startups, Portals and Investor Behavior*. London: Palgrave Macmillan. [CrossRef]
- Cumming, Douglas, and Sofia Johan. 2017. Crowdfunding and entrepreneurial internationalization. In *The World Scientific Reference on Entrepreneurship*. Singapore: The World Scientific Publishers, chp. 5, pp. 109–26. [CrossRef]
- Cumming, Douglas J., Gael Leboeuf, and Armin Schwienbacher. 2019. Crowdfunding Models: Keep-it-All versus All or Nothing. *Financial Management* forthcoming. [CrossRef]
- Cunningham, Katelan. 2016. With \$1.5M Success on Kickstarter, Hardware Startup Prynt Reinvents Photo Printing. Available online: http://thinkapps.com/blog/development/prynt-case-crowdfunded/ (accessed on 22 February 2020).
- Dixon, Huw David. 1992. The Competitive Outcome as the Equilibrium in an Edgeworthian Price-Quantity Model. *The Economic Journal* 102: 301–9. [CrossRef]
- Dixon, Huw David. 2001. Surfing Economics. Palgrave-MacMillan. Available online: https://www.amazon.co. uk/Surfing-Economics-Huw-David-Dixon/dp/033376062X (accessed on 22 February 2020).
- Dushnitsky, Gary, Massimiliano Guerini, Evila Piva, and Christina Rossi-Lamastra. 2016. Crowdfunding in Europe: Determinants of Platform Creation across Countries February. *California Management Review* 58: 44–71. [CrossRef]
- Dushnitsky, Gary, and Markus Fitza. 2018. Are We Missing the Platforms For the Crowd? Comparing Investment Drivers Across Multiple Crowdfunding Platforms. *Journal of Business Venturing Insights* 10: e00100. [CrossRef]
- Estrin, Saul, Daniel Gozman, and Susanna Khavul. 2018. The Evolution and Adoption of Equity Crowdfunding: Entrepreneur and Investor Entry into a New Market. *Small Business Economics* 51: 425–39. [CrossRef]
- Fairchild, Richard, Weixi Liu, and Yang Yao. 2017. An Entrepreneur's Choice of Crowd-Funding or Venture Capital Financing: The Effect of Entrepreneurial Overconfidence and CF-Investors' Passion. Working Paper. Available online: http://dx.doi.org/10.2139/ssrn.2926980 (accessed on 22 February 2020).
- Frydrych, Denis, Adam Bock, Tony Kinder, and Benjamin Koeck. 2014. Exploring entrepreneurial legitimacy in reward-based crowdfunding. *Journal Venture Capital An International Journal of Entrepreneurial Finance* 16: 247–69. [CrossRef]
- Fudenberg, Drew, and Jean Tirole. 1991. Game Theory. Cambridge: MIT Press. Available online: https://mitpress. mit.edu/books/game-theory (accessed on 22 February 2020).
- Gambardella, Massimiliano. 2012. How to (crowd-) fund and manage the (user-) innovation: The case of Big Buck Bunny. Paper presented at Workshop on Open Source and Design of Communication, Lisboa, Portugal, June 11.
- Gerber, Elizabeth, Julie Hui, and Pei-Yi Kuo. 2012. Crowdfunding: Why People are Motivated to Post and Fund Projects on Crowdfunding Platforms. Conference: Computer Supported Cooperative Work 2012. Available online: https://pdfs.semanticscholar.org/c1e2/a1068f0af1c3120c62be5943340518860ecb.pdf (accessed on 22 February 2020).
- Gleasure, Rob, and Joseph Feller. 2016. Emerging technologies and the democratisation of financial services: A metatriangulation of crowdfunding research. *Information and Organization* 26: 101–15. [CrossRef]
- Gleasure, Rob. 2015. Resistance to crowdfunding among entrepreneurs: An impression management perspective. *The Journal of Strategic Information Systems* 224: 219–33. [CrossRef]
- Greenberg, Michael, and Elizabeth Gerber. 2014. Learning to Fail: Experiencing Public Failure Online Through Crowdfunding. Paper presented at Conference Proceedings, 32nd Annual ACM Conference on Human Factors in Computing Systems, Toronto, ON, Canada, April 26–May 1.
- Harris, Milton, and Artur Raviv. 1991. The Theory of Capital Structure. *The Journal of Finance* 46: 297–355. [CrossRef]
- Hesse, Brendan. 2019. The Best Mobile Printers for Your Smartphone, Business Insider. Available online: https://www.businessinsider.com/best-mobile-printer?r=US&IR=T#the-best-for-printing-in-multiple-sizes-5 (accessed on 22 February 2020).
- Hildebrand, Thomas, Manju Puri, and Jörg Rocholl. 2016. Adverse Incentives in Crowdfunding. *Management Science* 63: 587–608. [CrossRef]

- Hughes, John S., and Jennifer L. Kao. 1997. Strategic Forward Contracting and Observability. *International Journal of Industrial Organization* 16: 121–33. [CrossRef]
- Jeitschko, Thomas, Ting Liu, and Tao Wang. 2016. Information Acquisition, Signaling and Learning in Duopoly. Working Papers from Stony Brook University, Department of Economics. Available online: https://ideas. repec.org/p/nys/sunysb/16-07.html (accessed on 22 February 2020).
- Kgoroeadira, Reabetswe, Andrew Burke, and André van Stel. 2019. Small Business Online Loan Crowdfunding: Who Gets Funded and What Determines the Rate of Interest? *Small Business Economics* 52: 67–87. [CrossRef]
- Kleinert, Simon, Christine Volkmann, and Marc Gr<sup>2</sup>unhagen, M. 2020. Third-party Signals in Equity Crowdfunding: The Role of Prior Financing. *Small Business Economics* 54: 341–65. [CrossRef]
- Kumar, Praveen, Nisan Landberg, and David Zvilichovsky. 2015. (Crowd)funding Innovation. Working Paper. Available online: https://www.aeaweb.org/conference/2016/retrieve.php?pdfid=661 (accessed on 22 February 2020).
- Kuo, Pei-Yi, and Elizabeth Gerber. 2012. Design principles: Crowdfunding as a creativity support tool. Paper presented at 2012 ACM Annual Conference on Human Factors in Computing Systems, Austin, TX, USA, May 5–10.
- Kuppuswamy, Venkat, and Barry Bayus. 2015. A Review of Crowdfunding Research and Findings. Handbook of New Product Development Research. Working Paper. Available online: http://dx.doi.org/10.2139/ssrn. 2685739 (accessed on 22 February 2020).
- Leland, Hayne, and David Pyle. 1977. Information Asymmetries, Financial Structure, and Financial Intermediation. Journal of Finance 32: 371–87. [CrossRef]
- Lenz, Rainer. 2016. Peer-to-Peer Lending—Opportunities and Risks. *European Journal of Risk and Regulation* 7: 688–700. [CrossRef]
- Li, Li, and Zixuan Wang. 2019. How does capital structure change product-market competitiveness? Evidence from Chinese firms. *PLoS ONE* 14: 1–14. [CrossRef] [PubMed]
- Llopis, Juan A. Sanchis, José María Millán, Rui Baptista, Andrew Burke, Simon Parker, and Roy Thurik. 2015. Good Times, Bad Times: Entrepreneurship and the Business Cycle. *International Entrepreneurship and Management Journal* 11: 243–51. [CrossRef]
- Lubit, Roy. 2001. Tacit Knowledge and Knowledge Management: The Keys to Sustainable Competitive Advantage. Organizational Dynamics 29. [CrossRef]
- Miglo, Anton. 2018. Crowdfunding in a Duopoly Under Asymmetric Information. Working Paper. Available online: https://mpra.ub.uni-muenchen.de/89016/1/MPRA\_paper\_89016.pdf (accessed on 22 February 2020).
- Miglo, Anton. 2020. Crowdfunding Under Market Feedback, Asymmetric Information And Overconfident Entrepreneur. Entrepreneurship Research Journal Forthcoming. Available online: https://mpra.ub.unimuenchen.de/89015/1/MPRA\_paper\_89015.pdf (accessed on 22 February 2020).
- Miglo, Anton, and Victor Miglo. 2019. Market Imperfections and Crowdfunding. *Small Business Economics Journal* 53: 51–79. [CrossRef]
- Mochkabadi, Kazem, and Christine Volkmann. 2018. Equity Crowdfunding: A Systematic Review of the Literature. Small Business Economics forthcoming. [CrossRef]
- Modigliani, Franco, and Merton Miller. 1958. The Cost of Capital, Corporation Finance And the Theory of Investment. *American Economic Review* 48: 261–97.
- Mogg, Trevor. 2017. Indiegogo Now Offers a Marketplace Where You Can Buy Its Coolest Stuff. Available online: https://www.digitaltrends.com/cool-tech/indiegogo-marketplace-online-store/ (accessed on 22 February 2020).
- Mollick, Ethan R. 2014. The Dynamics of Crowdfunding: An Exploratory Study. *Journal of Business Venturing* 29: 1–16. [CrossRef]
- Moritz, Alexandra, and Joern Hendrich Block. 2014. Crowdfunding: A Literature Review and Research Directions. Available online: http://ssrn.com/abstract=2554444 (accessed on 22 February 2020).
- Nachman, David, and Thomas Noe. 1994. Optimal Design of Securites under Asymmetric Information. *Review of Financial Studies* 7: 1–44. [CrossRef]
- Petruzzelli, Messeni, Angelo Natalicchio, Umberto Panniello, and Paolo Roma. 2019. Understanding the Crowdfunding Phenomenon and Its Implications for Sustainability. *Technological Forecasting & Social Change* 141: 138–48. [CrossRef]

- Rampini, Adriano. 2004. Entrepreneurial Activity, Risk, and the Business Cycle. *Journal of Monetary Economics* 51: 555–73. [CrossRef]
- Rear, Jack. 2018. Indiegogo's New Marketplace Wants to Help Crowdfunders Stand Out. Available online: https://www.verdict.co.uk/indiegogo-marketplace/ (accessed on 22 February 2020).
- Santos, Vasco. 2017. The Economics of Crowdfunding: Surplus Appropriation, and Allocative Efficiency'. Working Paper. Available online: https://editorialexpress.com/cgi-bin/conference/download.cgi?db\_ name=EEAESEM2017&paper\_id=2396 (accessed on 22 February 2020).
- Schiavone, Francesco. 2017. Incompetence and Managerial Problems Delaying Reward Delivery in Crowdfunding. Journal of Innovation Economics & Management 2: 185–207. [CrossRef]
- Schleifer, Theodore. 2017. Crowdfunding Site Indiegogo Will Now Let You Sell Products as It Shifts into Commerce. Available online: https://www.vox.com/2017/10/16/16474794/indiegogo-crowdfunding-commerce-amazon (accessed on 22 February 2020).
- Schwartz, Andrew. 2015. The Nonfinancial Returns of Crowdfunding. *Review of Banking and Financial Law* 34: 565–80.
- Singer, L., N. Seyff, and S. Fricker. 2011. Online social networks as a catalyst for software and IT innovation. Paper presented at 4th International Workshop on Social Software Engineering, Szeged, Hungary, September 5. [CrossRef]
- Smith, Craig. 2020. 15 Interesting Indiegogo Facts and Statistics (2020): By the Numbers. Available online: https://expandedramblings.com/index.php/indiegogo-facts-statistics/ (accessed on 22 February 2020).
- Strausz, Roland. 2017. Crowdfunding, Demand Uncertainty, and Moral Hazard—A Mechanism Design Approach. *American Economic Review* 107: 1430–76. [CrossRef]
- Sung-Min, Park. 2012. New business applications for social networking. SERI Quarterly 5: 121–25.
- Takahashi, Dean. 2018. Indiegogo Moves Beyond Crowdfunding to Help Startups with Manufacturing. Available online: https://venturebeat.com/2018/01/15/indiegogo-moves-beyond-crowdfunding-to-help-startups-with-manufacturing/ (accessed on 22 February 2020).
- Villot, Michel. 2011. Entrepreneurship and the Role of Risk: A Theoretical and Empirical Analysis of the Relationship between Entrepreneurship and Risk and the Differences Bet. Economics. Available online: http://hdl.handle.net/2105/9815 (accessed on 22 February 2020).
- Vismara, Silvio. 2016. Equity Retention and Social Network Theory in Equity Crowdfunding. *Small Business Economics* 46: 579–90. [CrossRef]
- Vismara, Silvio. 2018a. Information Cascades Among Investors in Equity Crowdfunding. *Entrepreneurship Theory* and Practice 42: 467–97. [CrossRef]
- Vismara, Silvio. 2018b. Signaling to Overcome Inefficiencies in Crowdfunding Markets. In *The Economics of Crowdfunding*. Amsterdam: Elsevier. [CrossRef]
- Vulkan, Nir, Thomas Åstebro, and Manuel Fernandez Sierra. 2016. Equity Crowdfunding: A new Phenomena. Journal of Business Venturing Insights 5: 37–49. [CrossRef]
- Wang, Yu-Lin, and Andrea Ellinger. 2011. Organizational Learning: Perception of External Environment and Innovation Performance. *International Journal of Manpower* 32: 512–36. [CrossRef]
- Williams, Joseph. 1995. Financial and industrial structure with agency. *The Review of Financial Studies* 8: 431–475. [CrossRef]
- Zvilichovsky, David, Yael Inbar, and Ohad Barzilay. 2013. Playing Both Sides of the Market: Success and Reciprocity on Crowdfunding Platforms. Economics, Computer Science. SSRN Electronic Journal 4. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).